### TRADEMARK ASSIGNMENT COVER SHEET

Electronic Version v1.1 Stylesheet Version v1.2 ETAS ID: TM868493

SUBMISSION TYPE:	NEW ASSIGNMENT	
NATURE OF CONVEYANCE:	Corrective assignment to correct the previously recorded assignment of App. No. 97826634 and Reg. Nos. 1553130, 1968491, 1971360, 3812456, 3816076, 4297249, and 4522990, which was recorded at REEL: 00820 and FRAME: 0125 through 0128.	

### **CONVEYING PARTY DATA**

Name	Formerly	Execution Date	Entity Type
Generations Coffee Company, LLC		01/11/2024	Corporation: DELAWARE

#### **RECEIVING PARTY DATA**

Name:	Generations Coffee Company, LLC	
Street Address:	10300 Brecksville Road	
City:	Brecksville	
State/Country: OHIO		
Postal Code: 44141		
Entity Type: Corporation: DELAWARE		

### **PROPERTY NUMBERS Total: 8**

Property Type	Number	Word Mark
Serial Number:	97826634	STEEP AND BREW
Registration Number:	1553130	STEEP & BREW
Registration Number:	1968491	WHOLE BEAN STEEP & BREW COFFEES
Registration Number:	1971360	WHOLE BEAN STEEP & BREW COFFEES
Registration Number:	3812456	COFFEE DONE RIGHT
Registration Number:	3816076	COFFEE DONE RIGHT
Registration Number:	4297249	CAFÉ FAIR
Registration Number:	4522990	KENDRA

### **CORRESPONDENCE DATA**

Fax Number: 2166871841

Correspondence will be sent to the e-mail address first; if that is unsuccessful, it will be sent using a fax number, if provided; if that is unsuccessful, it will be sent via US Mail.

Phone: 216-430-2205

Email: rwest@reminger.com **Correspondent Name:** Robert A. West, Jr. Address Line 1: 200 Public Square

**Address Line 2: Suite 1200** 

> **TRADEMARK** REEL: 008317 FRAME: 0551

900828387

Address Line 4:	Cleveland, OHIO 44114	
NAME OF SUBMITTER:	Robert A. West, Jr.	
SIGNATURE:	/Robert A. West, Jr./	
DATE SIGNED:	01/16/2024	

**Total Attachments: 59** source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page1.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page2.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page3.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page4.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page5.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page6.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page7.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page8.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page9.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page10.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page11.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page12.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page13.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page14.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page15.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page16.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page17.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page18.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page19.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page20.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page21.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page22.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page23.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page24.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page25.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page26.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page27.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page28.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page29.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page30.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page31.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page32.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page33.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page34.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page35.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page36.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page37.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page38.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page39.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page40.tif source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page41.tif

source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page42.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page43.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page44.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page45.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page46.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page47.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page48.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page49.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page50.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page51.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page52.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page53.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page54.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page55.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page56.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page57.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page58.tif
source=2024-01-11 - Affidavit of Dominic Caruso with Exhibits 1-3 (executed)#page59.tif

### IN THE UNITED STATES PATENT AND TRADEMARK OFFICE ASSIGNMENT RECORDATION BRANCH

In re:			
APPLICATION NUMBER:			
97826634			
REGISTRATION NUMBERS	3:		
1553130			
1968491			
1971360			
3812456 3816076			
4522990			
AFFID	AVIT	OF DOMINIC CARUSO	
STATE OF OHIO	)		
COUNTY OF CUYAHOGA	)	SS:	
Dominic Caruso, being first duly sv	worn an	nd under oath states as follows:	
1. I am of requisite age	and ca	apacity to give testimony in this matter.	
2. I have personal and	firsthan	nd knowledge of the facts stated in this Affidavit.	
3. I am the Chief Exec	utive O	officer for PMD Enterprises, Inc., d/b/a Caruso's	

TRADEMARK REEL: 008317 FRAME: 0554

members of Generations Coffee Company, LLC ("Generations Coffee"). See Limited Liability

Caruso's Coffee, along with Coffee Holding Co. ("Coffee Holding"), are the two

Coffee ("Caruso's Coffee).

Company Agreement of Generations Coffee Company, LLC (the "Agreement"), ¶1.12, 1.15, 1.39, Article V, execution page, and Exhibit A, a copy of which is attached hereto as Exhibit 1.

- 5. Generations Coffee's primary purpose is to roast, package, and sell private label specialty coffees to supermarkets nationwide, and to engage in any lawful act, as determined by the Board of Managers; its operations and the Board of Manager's binding authority are confined to actions necessary or convenient for the objectives outlined in Article III and under the constraints of Section 6.6 Restrictive Covenants of the Agreement. *Id.* at Article III.
- 6. No member shall have authority as a member to bind Generations Coffee. *Id.* at ¶6.1.
- 7. Generations Coffee shall not dispose of, purchase or lease any asset of Generations Coffee other than in the ordinary course of its business without the prior written consent of each member. *Id.* at ¶6.6(e).
- 8. The Board of Managers, comprising two managers elected by the members, retains exclusive authority over Generations Coffee's business affairs, each holding one vote on matters requiring Board of Managers approval; no manager can act for Generations Coffee without the other's written consent or a resolution of the Board of Managers, and decisions of the Board of Managers require a unanimous vote between the two managers. *Id.* at ¶7.1 and 7.2.
- 9. I, along with Andrew Gordon, make up the entirety of Generation Coffee's Board of Managers. *Id.* at Exhibit B.
- 10. On April 24, 2018, an assignment of trademarks was entered into by and between Generations Coffee, as assignee, and Steep & Brew, Inc., as assignor (the "2018 Assignment").

  See Trademark Reel 006589, Frame 0348, a copy of the Trademark Assignment Cover Sheet

with 2018 Assignment, marked Trademark Reel 006589, Frame 0346-0353, is attached hereto as **Exhibit 2**.

- 11. Steep & Brew, Inc. agreed to transfer and assign to Generations Coffee all of Steep & Brew, Inc.'s right, title and interest in, to and under its registered and unregistered marks and/or trade names. *Id*.
- 12. On March 4, 2019, a trademark assignment was recorded at Reel 006589, Frame 0346 through 0353, assigning the following trademarks from Steep & Brew, Inc. to Generations Coffee:

	Registration Number	Identification/Description
1	1553130	STEEP & BREW
2	1892293	WE KNOW BEANS ABOUT COFFEE
3	1955449	TOASTED NUT
4	1960051	WE KNOW BEANS ABOUT COFFEE
5	1968491	WHOLE BEAN STEEP & BREW COFFEES
6	1971360	WHOLE BEAN STEEP & BREW COFFEES
7	2051356	COFFEE CRAFTSMEN
8	2082016	COFFEE CRAFTSMEN
9	2076657	WORLD CUP
10	2078262	WORLD CUP
11	2706420	WORLD CUP – HAWAII MEXICO GUATEMALA
12	2774681	TOASTED NUT
13	3812456	COFFEE DONE RIGHT
14	3816076	COFFEE DONE RIGHT
15	4086431	CLUB FUND ZONE
16	4297249	CAFÉ FAIR

17	4502625	K2 COFFEE
18	4522990	KENDRA

- 13. On March 7, 2023, Generations Coffee filed a TEAS Plus Application for the standard work mark: STEEP AND BREW. *See* US Serial No. 97826634.
- 14. On November 27, 2023, Andrew Gordon executed, purportedly on behalf of Generations Coffee, a trademark assignment, attempting to assign to Express Coffee LLC all of Generations Coffee's rights, title and interests in and to the following:

	Application Number	Identification/Description
1	97826634	STEEP AND BREW
	Registration Number	<u>Identification/Description</u>
1	1553130	STEEP & BREW
2	1968491	WHOLE BEAN STEEP & BREW COFFEES
3	1971360	WHOLE BEAN STEEP & BREW COFFEES
4	3812456	COFFEE DONE RIGHT
5	3816076	COFFEE DONE RIGHT
6	4297249	CAFÉ FAIR
7	4522990	KENDRA

(the "2023 Assignment"). See Trademark Reel 008280, Frame 0127-0128, a copy of the Trademark Assignment Cover Sheet with 2023 Assignment, marked Trademark Reel 008280, Frame 0125-0128, is attached hereto as **Exhibit 3**.

- On November 27, 2023, the 2023 Assignment was recorded at Reel 008280,
   Frame 0125 through 0128. *Id*.
- 16. On December 19, 2023, I first learned of the 2023 Assignment. At no time did I, on behalf of Caruso's Coffee, as a member of Generations Coffee, provide written consent to Coffee Holding, as the other member, for the disposition of Generations Coffee's intellectual

property. Further, as one of the two managers of the Board of Managers, I neither provided my written consent for Andrew Gordon, the other manager, to act for Generations Coffee, nor voted on a resolution for Andrew Gordon to independently act on behalf of Generations Coffee.

- 17. As a result, Andrew Gordon did not possess the requisite authority to execute the 2023 Assignment on behalf of Generations Coffee, trying to dispose of its intellectual property.
  - 18. Therefore, the 2023 Assignment is improper.
- 19. Generations Coffee is the correct owner, pursuant to the 2018 Assignment (Reel 006589, Frame 0346-0353) and the filing of Application No. 97826634, of the following:

	Application Number	Identification/Description
1	97826634	STEEP AND BREW
	Registration Number	Identification/Description
1	1553130	STEEP & BREW
2	1968491	WHOLE BEAN STEEP & BREW COFFEES
3	1971360	WHOLE BEAN STEEP & BREW COFFEES
4	3812456	COFFEE DONE RIGHT
5	3816076	COFFEE DONE RIGHT
6	4297249	CAFÉ FAIR
7	4522990	KENDRA

FURTHER AFFIANT SAYETH NAUGHT.

1-11-24

Pominic Caruso

STATE OF OHIO	)
COUNTY OF CUYAHOGA	) SS: )
SWORN TO AND SUBSCRIBED	BEFORE ME, a Notary Public, this // day of
January 2024.	

Name:

**NOTARY PUBLIC** 

(stamp or seal)



SOUHEIL FARAH Notary Public, State of Ohio Commission No. 2015-RE-536834 My Commission Expires July 6, 2025

## EXHIBIT 1

# LIMITED LIABILITY COMPANY AGREEMENT OF GENERATIONS COFFEE COMPANY, LLC

A LIMITED LIABILITY COMPANY
ORGANIZED UNDER THE LAWS OF
THE STATE OF DELAWARE

[TPW: NYLEGAL:481186.7] 19499-00010 04/07/2006 02:17 PM

### TABLE OF CONTENTS

1.1	ARTICL	Æ I DEFINITIONS	1
1.2	1.1	Act	1
1.3       Affiliate          1.4       Agreement          1.5       Annual Independent Audit          1.6       Assignee or "Transferee"          1.7       Bankrupt Member          1.8       Board of Managers          1.9       Business Day          1.10       Capital Account          1.11       Capital Contribution          1.12       Caruso's Coffee          1.13       Certificate of Formation          1.14       Code          1.15       Coffee Holding          1.16       Commitment          1.17       Company          1.18       Company Property          1.19       Confidential Information          1.20       Default Interest Rate          1.21       Delinquent Member          1.22       Disposition (Dispose)          1.23       Dissociation          1.24       Dissolution Event          1.25       Distribution <td< td=""><td></td><td>Additional Capital Contribution</td><td>1</td></td<>		Additional Capital Contribution	1
1.4       Agreement          1.5       Annual Independent Audit          1.6       Assignee or "Transferee"          1.7       Bankrupt Member          1.8       Board of Managers          1.9       Business Day          1.10       Capital Account          1.11       Capital Contribution          1.12       Caruso's Coffee          1.13       Certificate of Formation          1.14       Code          1.15       Coffee Holding          1.16       Commitment          1.17       Company          1.18       Company Property          1.19       Confidential Information          1.20       Default Interest Rate          1.21       Delinquent Member          1.22       Disposition (Dispose)          1.23       Dissociation          1.24       Dissolution Event          1.25       Distribution          1.26       Economic Interest			
1.5       Annual Independent Audit         1.6       Assignee or "Transferee"       1         1.7       Bankrupt Member       1         1.8       Board of Managers       1         1.9       Business Day       1         1.10       Capital Account       1         1.11       Capital Contribution       2         1.12       Caruso's Coffee       2         1.13       Certificate of Formation       2         1.14       Code       2         1.15       Coffice Holding       2         1.16       Commitment       2         1.17       Company       2         1.18       Company Property       2         1.19       Confidential Information       2         1.20       Default Interest Rate       2         1.21       Delinquent Member       2         1.22       Dissociation (Dispose)       2         1.23       Dissociation (Dispose)       2         1.24       Dissolution Event       2         1.25       Distribution       2         1.26       Economic Interest       2         1.27       Effective Date       3         <			
1.6       Assignee or "Transferee"       1         1.7       Bankrupt Member       1         1.8       Board of Managers       1         1.9       Business Day       1         1.10       Capital Account       1         1.11       Capital Contribution       2         1.12       Caruso's Coffee       2         1.13       Certificate of Formation       2         1.14       Code       2         1.15       Coffee Holding       2         1.16       Commitment       2         1.17       Company       2         1.18       Company Property       2         1.19       Confidential Information       2         1.20       Default Interest Rate       2         1.21       Delinquent Member       2         1.22       Disposition (Dispose)       2         1.23       Dissociation       2         1.24       Dissociation       2         1.25       Distribution       2         1.26       Economic Interest       2         1.27       Effective Date       3         1.28       Equity Offering       3         1.29<			
1.7       Bankrupt Member          1.8       Board of Managers          1.9       Business Day          1.10       Capital Account          1.11       Cartiso's Coffee          1.12       Cartificate of Formation          1.13       Certificate of Formation          1.14       Code          1.15       Coffee Holding          1.16       Commitment          1.17       Company Property          1.18       Company Property          1.19       Confidential Information          1.20       Default Interest Rate          1.21       Delinquent Member          1.22       Disposition (Dispose)          1.23       Dissociation          1.24       Dissolution Event          1.25       Distribution          1.26       Economic Interest          1.27       Effective Date          1.28       Equity Offering          1.29       Exhibit A		•	
1.8       Board of Managers       1         1.9       Business Day       1         1.10       Capital Account       1         1.11       Capital Contribution       2         1.12       Caruso's Coffee       2         1.13       Certificate of Formation       2         1.14       Code       2         1.15       Coffee Holding       2         1.16       Commitment       2         1.17       Company       2         1.18       Company Property       2         1.19       Confidential Information       2         1.20       Default Interest Rate       2         1.21       Delinquent Member       2         1.22       Disposition (Dispose)       2         1.23       Dissociation       2         1.24       Dissociation       2         1.25       Distribution       2         1.26       Economic Interest       2         1.27       Effective Date       2         1.28       Equity Offering       3         1.29       Exhibit A       3         1.30       Fair Market Value       3         1.31			
1.9   Business Day		4	
1.10       Capital Account       1         1.11       Capital Contribution       2         1.12       Caruso's Coffee       2         1.13       Certificate of Formation       2         1.14       Code       2         1.15       Coffee Holding       2         1.16       Commitment       2         1.17       Company       2         1.18       Company Property       2         1.19       Confidential Information       2         1.20       Default Interest Rate       2         1.21       Delinquent Member       2         1.22       Dissosition (Dispose)       2         1.23       Dissosition (Dispose)       2         1.24       Dissolution Event       2         1.25       Distribution       2         1.26       Economic Interest       2         1.27       Effective Date       3         1.28       Equity Offering       3         1.29       Exhibit A       3         1.31       Fiscal Year       3         1.32       Gross Asset Value       3         1.33       Initial Capital Contribution       4			
1.11       Capital Contribution         1.12       Caruso's Coffee         1.13       Certificate of Formation         1.14       Code         1.15       Coffee Holding         1.16       Commitment         1.17       Company         1.18       Company Property         1.19       Confidential Information         1.20       Default Interest Rate         1.21       Delinquent Member         1.22       Disposition (Dispose)         1.23       Dissolution Event         1.24       Dissolution Event         1.25       Distribution         1.26       Economic Interest         1.27       Effective Date         1.28       Equity Offering         1.29       Exhibit A         1.30       Fair Market Value         1.31       Fiscal Year         1.32       Gross Asset Value         1.33       Initial Capital Contribution         1.34       Initial Wembership Interest         1.35       Initial Wembership Interest         1.36       Initial Whembership Interest         1.39       Member         40       Membership Interest         41<			
1.12       Caruso's Coffee       2         1.13       Certificate of Formation       2         1.14       Code       2         1.15       Coffee Holding       2         1.16       Commitment       2         1.17       Company       2         1.18       Company Property       2         1.19       Confidential Information       2         1.20       Default Interest Rate       2         1.21       Delinquent Member       2         1.22       Disposition (Dispose)       2         1.23       Dissolution Event       2         1.24       Dissolution Event       2         1.25       Distribution       2         1.26       Economic Interest       2         1.27       Effective Date       3         1.28       Equity Offering       3         1.29       Exhibit A       3         1.30       Fair Market Value       3         1.31       Fiscal Year       3         1.32       Gross Asset Value       3         1.33       Initial Membership Interest       4         1.34       Mitial Value       4         1			
1.13       Certificate of Formation       2         1.14       Code       2         1.15       Coffee Holding       2         1.16       Commitment       2         1.17       Company       2         1.18       Company Property       2         1.19       Confidential Information       2         1.20       Default Interest Rate       2         1.21       Delinquent Member       2         1.22       Disposition (Dispose)       2         1.23       Dissociation       2         1.24       Dissolution Event       2         1.25       Distribution       2         1.26       Economic Interest       2         1.27       Effective Date       3         1.28       Equity Offering       3         1.29       Exhibit A       3         1.30       Fair Market Value       3         1.31       Fiscal Year       3         1.32       Gross Asset Value       3         1.33       Initial Capital Contribution       4         1.34       Initial Membership Interest       4         1.35       Initial Management Right       4			
1.14       Code         1.15       Coffee Holding         1.16       Commitment         1.17       Company         1.18       Company Property         1.19       Confidential Information         1.20       Default Interest Rate         1.21       Delinquent Member         1.22       Disposition (Dispose)         1.23       Dissociation         1.24       Dissolution Event         1.25       Distribution         1.26       Economic Interest         1.27       Effective Date         1.28       Equity Offering         1.29       Exhibit A         1.30       Fair Market Value         1.31       Fiscal Year         1.32       Gross Asset Value         1.33       Initial Capital Contribution         1.34       Initial Membership Interest         1.35       Initial Membership Interest         1.36       Initial Value         1.37       Management Right         1.38       Manager         1.40       Membership Interest         1.41       Net Cash Flow         1.42       New Securities			
1.15       Coffee Holding       2         1.16       Commitment       2         1.17       Company       2         1.18       Company Property       2         1.19       Confidential Information       2         1.20       Default Interest Rate       2         1.21       Delinquent Member       2         1.22       Disposition (Dispose)       2         1.23       Dissociation       2         1.24       Dissolution Event       2         1.25       Distribution       2         1.26       Economic Interest       2         1.27       Effective Date       3         1.28       Equity Offering       3         1.29       Exhibit A       3         1.30       Fair Market Value       3         1.31       Fiscal Year       3         1.32       Gross Asset Value       3         1.33       Initial Capital Contribution       4         1.34       Initial Sharing Ratio       4         1.35       Initial Sharing Ratio       4         1.36       Initial Value       4         1.37       Manager       4 <t< td=""><td></td><td></td><td></td></t<>			
1.16       Commitment       2         1.17       Company       2         1.18       Company Property       2         1.19       Confidential Information       2         1.20       Default Interest Rate       2         1.21       Delinquent Member       2         1.22       Disposition (Dispose)       2         1.23       Dissociation       2         1.24       Dissolution Event       2         1.25       Distribution       2         1.26       Economic Interest       2         1.27       Effective Date       3         1.28       Equity Offering       3         1.29       Exhibit A       3         1.30       Fair Market Value       3         1.31       Fiscal Year       3         1.32       Gross Asset Value       3         1.33       Initial Capital Contribution       4         1.34       Initial Membership Interest       4         1.35       Initial Value       4         1.36       Initial Value       4         1.37       Management Right       4         1.38       Manager       4 <t< td=""><td></td><td>Coffee Holding</td><td>2</td></t<>		Coffee Holding	2
1.17       Company Property       2         1.18       Company Property       2         1.19       Confidential Information       2         1.20       Default Interest Rate       2         1.21       Delinquent Member       2         1.22       Disposition (Dispose)       2         1.23       Dissociation       2         1.24       Dissolution Event       2         1.25       Distribution       2         1.26       Economic Interest       2         1.27       Effective Date       3         1.28       Equity Offering       3         1.29       Exhibit A       3         1.30       Fair Market Value       3         1.31       Fiscal Year       3         1.32       Gross Asset Value       3         1.33       Initial Capital Contribution       4         1.34       Initial Membership Interest       4         1.35       Initial Value       4         1.36       Initial Value       4         1.37       Manager       4         1.38       Manager       4         1.40       Membership Interest       4			
1.18       Company Property         1.19       Confidential Information         1.20       Default Interest Rate         1.21       Delinquent Member         1.22       Disposition (Dispose)         1.23       Dissociation         1.24       Dissolution Event         1.25       Distribution         1.26       Economic Interest         1.27       Effective Date         1.28       Equity Offering         1.29       Exhibit A         1.30       Fair Market Value         1.31       Fiscal Year         1.32       Gross Asset Value         1.33       Initial Capital Contribution         1.34       Initial Membership Interest         1.35       Initial Sharing Ratio         1.36       Initial Value         1.37       Management Right         1.38       Manager         1.40       Membership Interest         1.41       Net Cash Flow         1.42       New Securities			
1.19       Confidential Information       2         1.20       Default Interest Rate       2         1.21       Delinquent Member       2         1.22       Disposition (Dispose)       2         1.23       Dissociation       2         1.24       Dissolution Event       2         1.25       Distribution       2         1.26       Economic Interest       2         1.27       Effective Date       3         1.28       Equity Offering       3         1.29       Exhibit A       3         1.30       Fair Market Value       3         1.31       Fiscal Year       3         1.32       Gross Asset Value       3         1.33       Initial Capital Contribution       4         1.34       Initial Membership Interest       4         1.35       Initial Sharing Ratio       4         1.36       Initial Value       4         1.37       Management Right       4         1.38       Manager       4         1.40       Membership Interest       4         1.41       Net Cash Flow       4         1.42       New Securities       4 </td <td>1.18</td> <td>Company Property</td> <td>2</td>	1.18	Company Property	2
1.20       Default Interest Rate	1.19		
1.21       Delinquent Member       2         1.22       Disposition (Dispose)       2         1.23       Dissociation       3         1.24       Dissolution Event       3         1.25       Distribution       2         1.26       Economic Interest       2         1.27       Effective Date       3         1.28       Equity Offering       3         1.29       Exhibit A       3         1.30       Fair Market Value       3         1.31       Fiscal Year       3         1.32       Gross Asset Value       3         1.33       Initial Capital Contribution       4         1.34       Initial Membership Interest       4         1.35       Initial Sharing Ratio       4         1.36       Initial Value       4         1.37       Management Right       4         1.38       Manager       4         1.39       Member       4         1.40       Membership Interest       4         1.41       Net Cash Flow       4         1.42       New Securities       4	1.20		
1.22       Disposition (Dispose)         1.23       Dissociation         1.24       Dissolution Event         1.25       Distribution         1.26       Economic Interest         1.27       Effective Date         1.28       Equity Offering         1.29       Exhibit A         1.30       Fair Market Value         1.31       Fiscal Year         1.32       Gross Asset Value         1.33       Initial Capital Contribution         4       1.34         Initial Membership Interest       4         1.35       Initial Sharing Ratio         4       4         1.36       Initial Value         4       4         1.38       Manager         4       4         1.39       Member         1.40       Membership Interest         1.41       Net Cash Flow         1.42       New Securities	1,21	Delinquent Member	2
1.23       Dissociation       2         1.24       Dissolution Event       2         1.25       Distribution       2         1.26       Economic Interest       2         1.27       Effective Date       3         1.28       Equity Offering       3         1.29       Exhibit A       3         1.30       Fair Market Value       3         1.31       Fiscal Year       3         1.32       Gross Asset Value       3         1.33       Initial Capital Contribution       4         1.34       Initial Membership Interest       4         1.35       Initial Sharing Ratio       4         1.36       Initial Value       4         1.37       Management Right       4         1.38       Manager       4         1.39       Member       4         1.40       Membership Interest       4         1.41       Net Cash Flow       4         1.42       New Securities       4	1.22		
1.24       Dissolution Event       2         1.25       Distribution       2         1.26       Economic Interest       2         1.27       Effective Date       3         1.28       Equity Offering       3         1.29       Exhibit A       3         1.30       Fair Market Value       3         1.31       Fiscal Year       3         1.32       Gross Asset Value       3         1.33       Initial Capital Contribution       4         1.34       Initial Membership Interest       4         1.35       Initial Sharing Ratio       4         1.36       Initial Value       4         1.37       Management Right       4         1.38       Manager       4         1.39       Member       4         1.40       Membership Interest       4         1.41       Net Cash Flow       4         1.42       New Securities       4			
1.25       Distribution	1.24		
1.26       Economic Interest	1.25		
1.27       Effective Date       3         1.28       Equity Offering       3         1.29       Exhibit A       3         1.30       Fair Market Value       3         1.31       Fiscal Year       3         1.32       Gross Asset Value       3         1.33       Initial Capital Contribution       4         1.34       Initial Membership Interest       4         1.35       Initial Sharing Ratio       4         1.36       Initial Value       4         1.37       Management Right       4         1.38       Manager       4         1.39       Member       4         1.40       Membership Interest       4         1.41       Net Cash Flow       4         1.42       New Securities       4	1.26		
1.28       Equity Offering       3         1.29       Exhibit A       3         1.30       Fair Market Value       3         1.31       Fiscal Year       3         1.32       Gross Asset Value       3         1.33       Initial Capital Contribution       4         1.34       Initial Membership Interest       4         1.35       Initial Sharing Ratio       4         1.36       Initial Value       4         1.37       Management Right       4         1.38       Manager       4         1.39       Member       4         1.40       Membership Interest       4         1.41       Net Cash Flow       4         1.42       New Securities       4	1.27		
1.29       Exhibit A       3         1.30       Fair Market Value       3         1.31       Fiscal Year       3         1.32       Gross Asset Value       3         1.33       Initial Capital Contribution       4         1.34       Initial Membership Interest       4         1.35       Initial Sharing Ratio       4         1.36       Initial Value       4         1.37       Management Right       4         1.38       Manager       4         1.39       Member       4         1.40       Membership Interest       4         1.41       Net Cash Flow       4         1.42       New Securities       4	1.28		
1.30       Fair Market Value       3         1.31       Fiscal Year       3         1.32       Gross Asset Value       3         1.33       Initial Capital Contribution       4         1.34       Initial Membership Interest       4         1.35       Initial Sharing Ratio       4         1.36       Initial Value       4         1.37       Management Right       4         1.38       Manager       4         1.39       Member       4         1.40       Membership Interest       4         1.41       Net Cash Flow       4         1.42       New Securities       4	1.29		
1.31       Fiscal Year	1.30		
1.32       Gross Asset Value       3         1.33       Initial Capital Contribution       4         1.34       Initial Membership Interest       4         1.35       Initial Sharing Ratio       4         1.36       Initial Value       4         1.37       Management Right       4         1.38       Manager       4         1.39       Member       4         1.40       Membership Interest       4         1.41       Net Cash Flow       4         1.42       New Securities       4	1.31		
1.33 Initial Capital Contribution       4         1.34 Initial Membership Interest       4         1.35 Initial Sharing Ratio       4         1.36 Initial Value       4         1.37 Management Right       4         1.38 Manager       4         1.39 Member       4         1.40 Membership Interest       4         1.41 Net Cash Flow       4         1.42 New Securities       4	1.32	Gross Asset Value	3
1.34 Initial Membership Interest       4         1.35 Initial Sharing Ratio       4         1.36 Initial Value       4         1.37 Management Right       4         1.38 Manager       4         1.39 Member       4         1.40 Membership Interest       4         1.41 Net Cash Flow       4         1.42 New Securities       4	1.33		
1.35       Initial Sharing Ratio       4         1.36       Initial Value       4         1.37       Management Right       4         1.38       Manager       4         1.39       Member       4         1.40       Membership Interest       4         1.41       Net Cash Flow       4         1.42       New Securities       4	1.34		
1.36       Initial Value       4         1.37       Management Right       4         1.38       Manager       4         1.39       Member       4         1.40       Membership Interest       4         1.41       Net Cash Flow       4         1.42       New Securities       4	1.35		
1.37       Management Right       4         1.38       Manager       4         1.39       Member       4         1.40       Membership Interest       4         1.41       Net Cash Flow       4         1.42       New Securities       4	1.36		
1.38 Manager       4         1.39 Member       4         1.40 Membership Interest       4         1.41 Net Cash Flow       4         1.42 New Securities       4	1.37		
1.39Member41.40Membership Interest41.41Net Cash Flow41.42New Securities4	1.38		
1.40Membership Interest	1.39		
1.41 Net Cash Flow4 1.42 New Securities4			
1.42 New Securities		•	
	1.43		

1.44	Officer	4
1.45	Organization	4
1.46	Person	5
1.47	Principal Office	5
1.48	Proceeding	5
1.49	Profits and Losses	5
1.50	Property	5
1.51	Regulations	5
1.52	Regulatory Allocations	6
1.53	Related Person	
1.54	Securities Act	6
1.55	Share	7
1.56	Sharing Ratio	
1.57	Substitute Member	
1.58	Tax Characterization and Additional Tax Terms	
1.59	Winding Up Sale	8
ARTICL	E II FORMATION	8
2.1		
2.1	Organization	
2.3	Agreement Name	
2.3 2.4	Term	
2.5	Registered Agent and Office	
2.6	Principal Office	
	•	
ARTICL	E III PURPOSE; NATURE OF BUSINESS	8
A D'TTCT	E IV ACCOUNTING AND RECORDS	n
4.1	Records to be Maintained	
4.2	Reports to Members	
4.3	Annual Audit	
4.4	Tax Returns and Reports	9
ARTICL	E V CLASSES OF MEMBERSHIP; NAMES AND ADDRESSES OF MEMBERS	9
A man combined and		_
ARTICL	E VI RIGHTS AND DUTIES OF MEMBERS; RESTRICTIVE COVENANTS	9
6.1	No Management Rights as Members	9
6.2	Liability of Members	
6.3	Indemnification	9
6.4	Representations, Warranties and Covenants	.10
6.5	Conflicts of Interest.	.10
6.6	Restrictive Covenants	
ARTICL	E VII BOARD OF MANAGERS AND OFFICERS	.11
7.1		
7.1	Board of Managers	
7.2 7.3	Voting	.11. + 1
1.5	Term of Office as a Member of the Board of Managers	. L L

7.4	Incapacitation of a Manager	12
7.5	Officers	11
7.6	Authority to Bind the Company	12
7.7	Actions of the Board of Managers	13
7.8	Indemnification	13
7.9	Resignation and Removal; Replacement	13
7.10	Other Activities	
7.11	Distributions	
7.12	Expenses	
7.13	Affiliates; Fees	
ARTIC	LE VIII CONTRIBUTIONS AND CAPITAL ACCOUNTS	14
8,1	Initial Capital Contributions	14
8.2	Indemnification	
8.3	Shares	
8.4	Additional Capital Contributions	
8.5	Enforcement of Commitments	
8.6	Capital Account	
8.7	No Obligation to Restore Deficit Balance	
8.8	Withdrawal; Successors	
8.9	Interest	
8.10	Investment of Capital Contributions and Company Cash	
8.11	Repayment of Capital Contribution.	16
ARTIC!	LE IX ALLOCATIONS AND DISTRIBUTIONS	
9.1	Profits and Losses	16
9.2	Profits	16
9.3	Special Allocations.	
9.4	Curative Allocations	
9.5	Section 754 Election	
9.6	Other Allocation Rules	
9.7	Distribution of Net Cash Flow.	
9.8	Allocation of Gain or Loss upon Winding Up.	21
ARTIC	LE X TAXES	21
10.1	Tax Matters Partner	21
ARTIC	LE XI TRANSFER OF MEMBERSHIP INTEREST	21
11.1	Compliance with Securities Laws and this Agreement	21
11.2	Transfer of Shares; Admission of Substitute Member	
11.3	Dispositions Not in Compliance with this Article Void	
11.4	Transfer to Affiliate of a Member	
11.5	Future Sales and Preemptive Rights	22
ARTIC:	LE XII DISSOCIATION OF A MEMBER	
12.1	Dissociation	23
12.2	Rights of Dissociating Member.	
	~	

ARTICLE XIII DISSOLUTION AND WINDING UP		24
13.1	Dissolution	24
13.2	Effect of Dissolution	24
13.3	Distribution of Assets on Dissolution	
13.4	Winding Up and Certificate of Dissolution	
ARTICL	E XIV MISCELLANEOUS	25
14.1	Termination of License	26
14.2	Notices	
14.3	Headings	
14.4	Entire Agreement	
14.5	Binding Agreement	25
14.6	Saving Clause	25
14.7	Counterparts	26
14.8	Governing Law	
14.9	No Partnership Intended for Nontax Purposes	26
14,10	No Rights of Creditors and Third Parties	26
14.11	•	
14.12	Confidentiality	
	Dispute Resolution	
	General Interpretive Principles	

### LIMITED LIABILITY COMPANY AGREEMENT OF GENERATIONS COFFEE COMPANY, LLC

This Limited Liability Company Agreement of Generations Coffee Company, LLC, a limited liability company organized pursuant to the Delaware Limited Liability Company Act, is entered into and shall be effective as of the Effective Date, by and among the Company and the persons executing this Agreement as Members.

### ARTICLE I DEFINITIONS

For purposes of this Agreement, unless the context clearly indicates otherwise, the following terms shall have the following meanings:

- 1.1 Act. The Delaware Limited Liability Company Act and all amendments thereto.
- 1.2 Additional Capital Contribution. An additional Capital Contribution payable by a Member to the Company pursuant to Article VIII.
- 1.3 Affiliate. With respect to any Person, any entity controlling, controlled by or under common control with such Person. "Control" (including, with correlative meanings, the terms "controlled by" and "under common control with"), as used with respect to any Person, shall mean the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of over 50% of the voting securities of such Person, by contract or otherwise.
- 1.4 Agreement. This Limited Liability Company Agreement including all amendments adopted in accordance with this Agreement and the Act.
- 1.5 Annual Independent Audit. Annual Independent Audit shall have the meaning set forth in Section 4.3.
- 1.6 Assignce or "Transferce." A transferce of an Economic Interest who has not been admitted as a Substitute Member. Unless otherwise clear from the context of its use, the term "transferce" is synonymous with the term "Assignee."
- 1.7 **Bankrupt Member.** A Member who: (a) files a petition in bankruptcy, or is adjudicated as bankrupt or insolvent, or makes an assignment for the benefit of creditors, or an arrangement pursuant to any bankruptcy law; (b) discontinues or dissolves its business; or (c) has a receiver appointed for such Member or such Member's business and such receiver is not discharged within sixty (60) days.
- 1.8 Board of Managers. A Board constituted with those Persons selected to manage the affairs of the Company under Article VII hereof.
- 1.9 **Business Day.** Any day other than Saturday, Sunday or any legal holiday observed in the State of Delaware.
- 1.10 Capital Account. The account maintained for a Member or an Assignee determined in accordance with Article VIII.

- 1.11 Capital Contribution. A Member's Initial Capital Contribution plus any Additional Capital Contribution made by the Member in accordance with this Agreement. A Capital Contribution includes (a) the amount of any money contributed by the Member to the Company (including liabilities of the Company assumed by the Member as provided in section 1.704-1(b)(2)(iv)(c) of the Regulations), and (b) the Gross Asset Value of any Property contributed to the Company by such Member (net of liabilities secured by such contributed Property that the Company is considered to assume or take subject to under section 752 of the Code).
- 1.12 Caruso's Coffee. PMD Enterprises, Inc. (doing business as Caruso's Coffee), an Ohio corporation.
- 1.13 Certificate of Formation. The Certificate of Formation of the Company, as amended from time to time, filed with the Secretary of State of the State of Delaware.
  - 1.14 Code. The Internal Revenue Code of 1986, as amended and in effect from time to time.
  - 1.15 Coffee Holding. Coffee Holding Co., Inc., a Nevada corporation.
- 1.16 Commitment. The Initial Capital Contribution and Additional Capital Contributions that a Member is obligated to make.
- 1.17 Company. Generations Coffee Company, LLC, a limited liability company formed under the laws of Delaware, and any successor limited liability company
  - 1.18 Company Property. Any Property owned by the Company.
- 1.19 Confidential Information. Confidential Information shall have the meaning set forth in Section 14.10.
- 1.20 **Default Interest Rate.** The prime rate published by the Wall Street Journal, New York City edition, for the last Business Day on which a Commitment is payable.
- 1.21 **Delinquent Member.** Delinquent Member shall have the meaning set forth in Section 8.4.
- 1.22 **Disposition (Dispose).** Any sale, assignment, exchange, mortgage, pledge, grant, hypothecation or other transfer, absolute or as security or encumbrance (including dispositions by operation of law).
- 1.23 **Dissociation.** Any action which causes a Person to cease to be a Member as described in Article XIII hereof.
- 1.24 **Dissolution Event.** An event, the occurrence of which will result in the dissolution of the Company under Article XIII.
  - 1.25 **Distribution.** A transfer of Property to a Member on account of a Membership Interest.
- 1.26 **Economic Interest.** The right to receive allocations of Profits and Losses, Distributions, returns of capital and distribution of assets upon a dissolution of the Company.

- 1.27 Effective Date. April 7, 2006.
- 1.28 Equity Offering. Equity Offering shall have the meaning set forth in Section 11.5.
- 1.29 Exhibit A. Exhibit A to this Agreement setting forth the name, address, Initial Capital Contribution, Initial Membership Interest and Initial Sharing Ratio of each Member.

### 1.30 Fair Market Value.

- (a) As of any date, the fair market value of an asset on such date as determined in good faith by the Board of Managers. For this purpose, the Board of Managers may in its reasonable and prudent discretion value assets that are restricted by law, contract, market conditions (including trading volume relative to the Company's holding) or otherwise as to salability or transferability at an appropriate discount, based on the nature and term of such restrictions.
- (b) With respect to any Membership Interest or Shares, fair market value shall be based on the Company being operated as a going concern without any adjustment for marketability or minority interest. Fair market value shall be determined, in accordance with the preceding sentence, in good faith by the Board of Managers.
  - 1.31 Fiscal Year. The twelve month period ending on October 31 of each calendar year.
- 1.32 Gross Asset Value. Gross Asset Value, with respect to any Company asset, means the adjusted basis of such asset for federal income tax purposes, except as follows:
  - (a) The initial Gross Asset Value of any Company asset contributed by a Member to the Company shall be the Fair Market Value of such Company asset as of the date of such contribution;
  - (b) The Gross Asset Value of each Company asset shall be adjusted to equal its Fair Market Value, as of the following times: (i) the acquisition of an additional Membership Interest by any new or existing Member in exchange for more than a de minimis Capital Contribution unless the Board of Managers determines that such adjustment is not necessary to reflect the relative Economic Interests of the Members in the Company; (ii) the Distribution by the Company to a Member of more than a de minimis amount of Company assets (other than cash) as consideration for all or part of its Membership Interest unless the Board of Managers determines that such adjustment is not necessary to reflect the relative Economic Interests of the Members in the Company; and (iii) the liquidation of the Company within the meaning of section 1.704-1(b)(2)(ii)(g) of the Regulations;
  - (c) The Gross Asset Value of a Company asset distributed to any Member shall be the Fair Market Value of such Company asset as of the date of Distribution thereof;
  - (d) The Gross Asset Value of each Company asset shall be increased or decreased, as the case may be, to reflect any adjustments to the adjusted basis of such Company asset pursuant to section 734(b) or 743(b) of the Code, but only to the extent that such adjustments are taken into account in determining Capital Accounts pursuant to section 1.704-1(b)(2)(iv)(m) of the Regulations; and

- (e) If the Gross Asset Value of a Company asset has been determined or adjusted pursuant to paragraph (a), (b) or (d) above, such Gross Asset Value shall thereafter be adjusted to reflect the depreciation or amortization taken into account with respect to such Company asset for purposes of computing Profits and Losses.
- 1.33 Initial Capital Contribution. With respect to any Member, the Initial Capital Contribution set forth in Exhibit A opposite the name of such Member.
- 1.34 Initial Membership Interest. With respect to any Member, the Initial Membership Interest (as represented in Shares) set forth in Exhibit A opposite the name of such Member.
- 1.35 Initial Sharing Ratio. With respect to any Member, the Initial Sharing Ratio set forth in Exhibit A opposite the name of such Member.
- 1.36 Initial Value. With respect to any Member, the Initial Value set forth on Exhibit A opposite the name of such Member.
- 1.37 Management Right. The right, if any, of a Member to participate in the management of the Company, to vote on any matter and to grant or withhold consent or approval of actions of the Company.
  - 1.38 Manager. Manager shall have the meaning set forth in Section 7.1.
  - 1.39 Member. A party executing this Agreement and a Substitute Member.
- 1.40 Membership Interest. A Member's Economic Interest and Management Right which shall be represented by Shares of the Company.
- 1.41 Net Cash Flow. Net Cash Flow shall mean, with respect to any fiscal period of the Company, all cash revenues of the Company during that period, decreased by, without duplication, (a) cash expenditures for operating expenses, (b) capital expenditures to the extent not made from reserves, (c) repayment of principal on any financing and (d) taxes.
- 1.42 New Securities. Means any equity interests in the Company issued after the date of this Agreement whether now authorized or not, including, but not limited to, Shares, Membership Interests or any warrants, options, convertible securities or other contracts or rights to purchase Shares, Membership Interests or securities of the Company.
- 1.43 Offering Allocation. Offering Allocation shall have the meaning set forth in Section 11.5.
- 1.44 Officer. Any individual appointed as an officer of the Company in accordance with Section 7.4.
- 1.45 Organization. A Person other than a natural person. Organization includes, without limitation, corporations (both non-profit and other corporations), partnerships (both limited and general), joint ventures, limited liability companies and unincorporated associations, but the term does not include joint tenancies and tenancies by the entirety.

- 1.46 **Person.** An individual, trust, estate or any Organization permitted to be a member of a limited liability company under the laws of the State of Delaware.
  - 1.47 **Principal Office.** The Principal Office of the Company set forth in Section 2.6.
- 1.48 **Proceeding.** Any administrative, judicial or other adversary proceeding, including, without limitation, litigation, arbitration, administrative adjudication, mediation and appeal or review of any of the foregoing.
- 1.49 **Profits and Losses.** For each Fiscal Year or other period, an amount equal to the Company's taxable income or loss for such Fiscal Year or period, determined in accordance with section 703(a) of the Code (for this purpose, all items of income, gain, loss or deduction required to be stated separately pursuant to section 703(a)(1) of the Code shall be included in taxable income or loss), with the following adjustments:
  - (a) Any income of the Company that is exempt from federal income tax and not otherwise taken into account in computing Profits or Losses pursuant to this Section shall be added to such taxable income or loss;
  - (b) Any expenditures of the Company described in section 705(a)(2)(B) of the Code or treated as section 705(a)(2)(B) of the Code expenditures pursuant to section 1.704-1(b)(2)(iv)(i) of the Regulations (other than expenses in respect of which an election is properly made under section 709 of the Code), and not otherwise taken into account in computing Profits or Losses pursuant to this Section, shall be subtracted from such taxable income or loss;
  - (c) In the event the Gross Asset Value of any Company asset is adjusted pursuant to paragraph (b) or (d) of the definition of Gross Asset Value, the amount of such adjustment shall be taken into account as gain or loss from the disposition of such Company asset for purposes of computing Profits or Losses;
  - (d) Gain or loss resulting from any disposition of any Company asset with respect to which gain or loss is recognized for federal income tax purposes shall be computed by reference to the Gross Asset Value of the Company asset disposed of, notwithstanding that the adjusted tax basis of such Company asset may differ from its Gross Asset Value;
  - (e) In accordance with section 1.704-1(b)(2)(iv)(g)(3) of the Regulations, depreciation with respect to any Company asset shall be computed by reference to the adjusted Gross Asset Value of such asset, notwithstanding that the adjusted tax basis of such Company asset differs from its Gross Asset Value; and
  - (f) Notwithstanding any other provisions of this definition, Profits and Losses shall not include allocations under section 704(c) (which are set forth at Section 9.3(i) hereof) or Regulatory Allocations.
- 1.50 **Property.** Any property, real or personal, tangible or intangible, including money, and any legal or equitable interest in such property, but excluding services and promises to perform services in the future.
- 1.51 **Regulations.** The final and temporary federal income tax regulations promulgated by the United States Treasury Department under the Code as such regulations may be amended from time to

time, or if no final or temporary regulations with respect to a tax issue are then in effect, proposed regulations then in effect if approved by the Board of Managers. All references herein to a specific section of the Regulations shall be deemed also to refer to any corresponding provision of succeeding Regulations.

1.52 **Regulatory Allocations.** Regulatory Allocations shall mean the allocations set forth at Sections 9.3(a) through (g).

#### 1.53 Related Person. Means:

With respect to a particular individual:

- (a) each member of such individual's Family;
- (b) any Person that is directly or indirectly controlled by any one or more members of such individual's Family;
- (c) any Person in which members of such individual's Family hold (individually or in the aggregate) a Material Interest; and
- (d) any Person with respect to which one or more members of such individual's Family serves as a director, officer, partner, executor or trustee (or in a similar capacity).

With respect to a specified Person other than an individual:

- (a) any Person that directly or indirectly controls, is directly or indirectly controlled by or is directly or indirectly under common control with such specified Person;
  - (b) any Person that holds a Material Interest in such specified Person;
- (c) each Person that serves as a director, officer, partner, executor or trustee of such specified Person (or in a similar capacity);
  - (d) any Person in which such specified Person holds a Material Interest; and
- (e) any Person with respect to which such specified Person serves as a general partner or a trustee (or in a similar capacity).

For purposes of this definition, (a) "control" (including "controlling," "controlled by," and "under common control with") means the possession, direct or indirect, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract or otherwise, and shall be construed as such term is used in the rules promulgated under the Securities Act; (b) the "Family" of an individual includes (i) the individual, (ii) the individual's spouse, (iii) any other natural person who is related to the individual or the individual's spouse within the second degree and (iv) any other natural person who resides with such individual; and (c) "Material Interest" means direct or indirect beneficial ownership (as defined in Rule 13d-3 under the Securities Exchange Act of 1934) of voting securities or other voting interests representing at least ten percent (10%) of the outstanding voting power of a Person or equity securities or other equity interests representing at least ten percent (10%) of the outstanding equity securities or equity interests in a Person.

1.54 Securities Act. The Securities Act of 1933, as amended.

- 1.55 Share. A share of a Membership Interest as described in Section 8.2.
- 1.56 Sharing Ratio. With respect to any Member as of any date, the ratio (expressed as a percentage) of (a) such Member's Shares to (b) the aggregate outstanding Shares of all Members, or such other ratio as shall be agreed by all Members from time to time. The Initial Membership Interest and the Initial Sharing Ratio of each Member are set forth in Exhibit A, and Exhibit A shall be amended as necessary to conform to any changes thereto agreed to by the Members. In the event all or any portion of a Membership Interest is transferred in accordance with the terms of this Agreement, the transferre shall succeed to the Membership Interest and Sharing Ratio of the transferor to the extent it relates to the transferred Membership Interest.
- 1.57 Substitute Member. An Assignee who has been admitted to all of the rights of membership pursuant to Section 11.2.
- 1.58 Tax Characterization and Additional Tax Terms. It is intended that the Company be characterized and treated as a partnership for, and solely for, federal, state and local income tax purposes. For such purpose, the Company shall be subject to all of the provisions of subchapter K of chapter 1 of subtitle A of the Code, and all references to a "Partner," to "Partners" and to the "Partnership" in this Agreement (including the provisions of Articles VIII and IX) and in the provisions of the Code and Regulations cited in this Agreement shall be deemed to refer to a Member, the Members and the Company, respectively. In addition, the following terms shall have the following meanings:
  - (a) Adjusted Capital Account Deficit shall mean, with respect to any Member, the deficit balance, if any, in such Member's Capital Account as of the end of the relevant Fiscal Year, after giving effect to the following adjustments:
    - (i) Credit to such Capital Account any amounts which such Member is deemed to be obligated to restore pursuant to section 1.704-1(b)(2)(ii)(c) or the penultimate sentences of sections 1.704-2(g)(1) and 1.704-2(i)(5) of the Regulations; and
    - (ii) Debit to such Capital Account the items described in sections 1.704-1(b)(2)(ii)(d)(4), 1.704-1(b)(2)(ii)(d)(5) and 1.704-1(b)(2)(ii)(d)(6) of the Regulations.

The foregoing definition of Adjusted Capital Account Deficit is intended to comply with the provisions of section 1.704-1(b)(2)(ii)(d) of the Regulations and shall be interpreted consistently therewith.

- (b) <u>Nonrecourse Deductions</u> has the meaning set forth in section 1.704-2(b)(1) of the Regulations.
- (c) Nonrecourse Liability has the meaning set forth in section 1.704-2(b)(3) of the Regulations.
- (d) <u>Partner Nonrecourse Debt</u> has the meaning set forth in section 1.704-2(b)(4) of the Regulations.
- (e) <u>Partner Nonrecourse Debt Minimum Gain</u> means an amount, with respect to each Partner Nonrecourse Debt, equal to the Partnership Minimum Gain that would result if such Partner Nonrecourse Debt were treated as a Nonrecourse Liability, determined in accordance with section 1.704-2(i)(3) of the Regulations.

- (f) <u>Partner Nonrecourse Deductions</u> has the meaning set forth in sections 1.704-2(i)(1) and 1.704-2(i)(2) of the Regulations.
- (g) <u>Partnership Minimum Gain</u> has the meaning set forth in sections 1.704-2(b)(2) and 1.704-2(d) of the Regulations.
- 1.59 Winding Up Sale. Winding Up Sale shall have the meaning set forth in Section 9.9(a).

### ARTICLE II FORMATION

- 2.1 **Organization.** The Members hereby organize the Company as a Delaware limited liability company pursuant to the provisions of the Act.
- 2.2 Agreement. For and in consideration of the mutual covenants herein contained and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Members executing this Agreement hereby agree to the terms and conditions of this Agreement, as it may from time to time be amended. It is the express intention of the Members that this Agreement shall be the sole source of agreement of the parties with respect to the subject matter hereof, and, except to the extent a provision of this Agreement expressly incorporates federal income tax rules by reference to sections of the Code or Regulations or is expressly prohibited or ineffective under the Act, this Agreement shall govern, even when inconsistent with, or different than, the provisions of the Act or any other law or rule. To the extent any provision of this Agreement is prohibited or ineffective under the Act, this Agreement shall be considered amended to the smallest degree possible in order to make this Agreement effective under the Act. In the event the Act is subsequently amended or interpreted in such a way to make any provision of this Agreement that was formerly invalid valid, such provision shall be considered to be valid from the effective date of such interpretation or amendment.
- 2.3 Name. The name of the Company is Generations Coffee Company, LLC, and all business of the Company shall be conducted under that name or under any other name determined by the Board of Managers but, in any case, only to the extent permitted by applicable law.
- 2.4 **Term.** The term of the Company shall be perpetual unless the Company shall be sooner dissolved and its affairs wound up in accordance with the Act or this Agreement.
- 2.5 Registered Agent and Office. The registered agent for the service of process and the registered office shall be that Person and location reflected in the Certificate of Formation as filed in the office of the Secretary of State of the State of Delaware. The Board of Managers may, from time to time, change the registered agent or office through appropriate filings with the Secretary of State of the State of Delaware. In the event the registered agent ceases to act as such for any reason or the registered office shall change, the Board of Managers shall promptly designate a replacement registered agent or file a notice of change of address, as the case may be. If the Board of Managers shall fail to designate a replacement registered agent or change of address of the registered office, any Member may designate a replacement registered agent or file a notice of change of address.
- 2.6 **Principal Office.** The Principal Office of the Company shall be located at 10300 Brecksville, Road, Brecksville, OH 44141.

### ARTICLE III PURPOSE; NATURE OF BUSINESS

The purpose of the Company is to engage in the roasting, packaging and sale of private label specialty coffees for sale and distribution to supermarkets throughout the United States, and to engage in any other lawful act or activity, as determined by the Board of Managers from time to time, for which limited liability companies may be organized under the Act. The Company shall have the authority to do all things necessary or convenient to accomplish its purposes and operate its business as described in this Article III. The Company exists only for the purposes specified in this Article III. The authority granted to the Board of Managers under this Agreement to bind the Company shall be limited to actions necessary or convenient to the purposes specified in this Article III and to the extent provided by Section 6.6.

### ARTICLE IV ACCOUNTING AND RECORDS

- 4.1 Records to be Maintained. The Company shall maintain at the Principal Office, the Company's books and records, including financial statements of the Company and all other documents and data regarding the Company, which shall be open to inspections by the Members or their agents at any reasonable time.
- 4.2 **Reports to Members.** The Board of Managers shall provide reports to the Members, including a balance sheet, a statement of profit and loss and changes in Members' accounts and a statement of cash flows, within fifteen (15) days of the end of each month.
- 4.3 Annual Audit. Within ninety (90) days of the end of each Fiscal Year, the Board of Managers shall cause the annual financial statements of the Company to be audited by the independent registered public accounting firm responsible for conducting the audit of the financial statements of Coffee Holding for the same period ("Annual Independent Audit").
- 4.4 Tax Returns and Reports. The Board of Managers, at the Company's expense, shall prepare and timely file income tax returns of the Company in all jurisdictions where such filings are required, and shall prepare and deliver to each Member, within the time prescribed by the Code and any extensions applicable thereto, as provided by the Code or applicable regulations, all information returns required by the Code and Company information necessary for the preparation of the Members' federal income tax returns.

### ARTICLE V CLASSES OF MEMBERSHIP; NAMES AND ADDRESSES OF MEMBERS

There shall be only one class of Members. The names and addresses of the Members are as stated on Exhibit A.

### ARTICLE VI RIGHTS AND DUTIES OF MEMBERS; RESTRICTIVE COVENANTS

- 6.1 No Management Rights as Members. No Member shall have authority as a Member to bind the Company.
- 6.2 **Liability of Members.** No Member shall be liable as such for the debts, obligations or liabilities of the Company.
- 6.3 **Indemnification.** A Member shall indemnify the Company for any costs or damages incurred by the Company as a result of any unauthorized action by such Member.

9

6.4 Representations, Warranties and Covenants. Each Member, and in the case of a trust or other Organization, the person(s) executing this Agreement on behalf of such trust or other Organization, hereby represents and warrants to the Company and to each other Member that: (a) if that Member is a trust or other Organization, it has the requisite power to enter into this Agreement and to perform its obligations hereunder and that the person(s) executing this Agreement on behalf of such trust or other Organization has the requisite power to do so; and (b) the Member is acquiring its interest in the Company for the Member's own account as an investment and without an intent to distribute the interest. The Members acknowledge that their interests in the Company have not been registered under the Securities Act or any state securities laws and may not be resold or transferred without appropriate registration or the availability of an exemption from such requirements.

#### 6.5 Conflicts of Interest.

- (a) The Members shall account to the Company and hold as trustee for it any Property, Profit or benefit derived by the Member, without the consent of all of the other Members, in the conduct and winding up of the Company business or from a use or appropriation by the Member of Company Property, including information developed exclusively for the Company and opportunities expressly offered to the Company.
- (b) A Member does not violate a duty or obligation to the Company merely because the Member's conduct furthers the Member's own interest. A Member may lend money to and transact other business with the Company. The rights and obligations of a Member who lends money to or transacts business with the Company are the same as those of a Person who is not a Member, subject to other applicable law. No transaction with the Company shall be voidable solely because a Member has a direct or indirect interest in the transaction if the transaction is fair to the Company.
- 6.6 **Restrictive Covenants.** Notwithstanding anything contained herein to the contrary, the Company shall not take any of the following actions without the prior written consent of each Member:
  - (a) amend the Certificate of Formation or this Agreement;
  - (b) purchase any interest in the stock, assets or business of any corporation, partnership or other entity other than in the ordinary course of the Company's business;
  - (c) appoint or discharge any members of the Board of Managers of the Company, other than pursuant to the provisions set forth in Article VII;
  - (d) enter into any arrangement which will result in the merger, consolidation, dissolution, liquidation, winding up or cessation of the Company or the business activities of the Company;
  - (e) dispose of, purchase or lease any asset of the Company other than in the ordinary course of the Company's business;
    - (f) issue any New Securities;
    - (g) require Additional Capital Contributions from the Members;
    - (h) admit any new Members or permit the withdrawal of any existing Member;

- (i) file a voluntary petition or otherwise initiate proceedings to have the Company adjudicated bankrupt or insolvent, or consent to the institution of bankruptcy or insolvency proceedings against the Company, or file a petition seeking or consenting to reorganization or relief of the Company as debtor under any applicable federal or state law relating to bankruptcy, insolvency, or other relief for debtors with respect to the Company; or seek or consent to the appointment of any trustee, receiver, conservator, assignee, sequestrator, custodian, liquidator (or other similar official) of the Company or of all or any substantial part of the properties and assets of the Company, or make any general assignment for the benefit of creditors of the Company, or admit in writing the inability of Company to pay its debts generally as they become due or declare or effect a moratorium on the Company debt or take any action in furtherance of any such action;
- (j) sell or transfer any asset of the Company to a Related Person unless such sale or transfer is made at Fair Market Value;
- (k) increase or decrease the number of Managers the Company is authorized to have; or
  - (1) authorize additional Shares.

### ARTICLE VII BOARD OF MANAGERS AND OFFICERS

- 7.1 Board of Managers. The management of the Company and all decisions concerning the business affairs of the Company shall be made by the Board of Managers. The Board of Managers shall consist of two (2) managers, one of which shall be elected by each of the Members. Each Manager shall have one (1) vote with respect to all matters requiring the vote of the Board of Managers. No individual member of the Board of Managers (a "Manager") shall have the right to act on behalf of the Company without the written consent of the other Manager or absent authorization by resolution of the Board of Managers. The initial Board of Managers shall consist of those persons listed on Exhibit B hereto.
- 7.2 Voting. Except as otherwise provided in this Agreement, the affirmative vote of a majority of the members of the Board of Managers shall constitute the act of the Board of Managers. If there are only two members on the Board of Managers, then the affirmative vote of both members shall constitute the act of the Board of Managers.
- 7.3 Term of Office as a Member of the Board of Managers. Each member of the Board of Managers shall serve until his or her removal by the Member who appointed him or her or any resignation of such member of the Board of Managers pursuant to Section 7.8. A member of the Board of Managers may be removed by the Member who appointed him or her at any time, with or without cause.
- 7.4 Incapacitation of a Manager. In the event that any Manager becomes incapacitated such that he or she us incapable in any material respect of performing the services required by such Manager in accordance with this Agreement on a regular ongoing basis for a period of sixty (60) days, such Manager shall automatically be removed from the Board of Managers and replaced by a newly elected Manager as shall be determined by the Member who appointed such incapacitated Manager to the Board of Managers.

- 7.5 Officers. The Board of Managers may appoint one or more Officers of the Company as it shall determine from time to time advisable with such titles as the Board of Managers deems appropriate. Any two or more offices may be held by the same person. All Officers shall have such authority and perform such duties in the management and operation of the Company as may be prescribed by the Board of Managers or this Agreement. Any Officer may be removed at any time, with or without cause, by the Board of Managers, and any vacancy occurring in any office shall be filled by the Board of Managers.
- 7.6 Authority to Bind the Company. The Officers shall have such authority as is granted by the Board of Managers (including, if authorized, the authority to bind the Company), subject to the authority of the Board of Managers. The Board of Managers has the power, on behalf of the Company, to do all things necessary or convenient to carry out the business and affairs of the Company (as described in Article III), including, without limitation:
  - (a) To institute, prosecute and defend any Proceeding in the Company's name;
  - (b) To conduct the Company's business, establish Company offices and exercise the powers of the Company;
  - (c) To employ, contract and deal with, from time to time, Persons, including any Member or Affiliate of any Member, in connection with the management and operation of the Company's business, including without limitation, suppliers, customers, tradespeople, brokers, accountants and attorneys, on such terms as the Board of Managers shall determine;
  - (d) To purchase liability and other insurance to protect the Company's business and Property;
  - (e) To establish reserve funds of the Company to provide for future requirements for operations, contingencies or any other purpose that the Board of Managers deems necessary or appropriate;
  - (f) To make such elections under the Code and other relevant tax laws as to the treatment of items of Company income, gain, loss, deduction and credit, and as to all other relevant matters as the Board of Managers, with the advice of the Company's accountants and attorneys, deems necessary or appropriate, including without limitation, elections referred to in section 754 of the Code (subject to Section 9.6), the determination of which items of cash outlay shall be capitalized or treated as current expenses, and the selection of the method of accounting and bookkeeping procedures to be used by the Company;
  - (g) To pay as a Company expense any and all costs or expenses associated with the formation, development, organization and operation of the Company;
  - (h) To deposit, withdraw, invest, pay, retain and distribute the Company's funds in a manner consistent with the provisions of this Agreement; and
  - (i) To execute, acknowledge and deliver any and all instruments to effectuate the foregoing.

All Company cash shall be deposited in a bank account selected by the Board of Managers, and all disbursements of Company cash shall be approved in advance by those Persons designated by the Board of Managers,

- 7.7 Actions of the Board of Managers. The Board of Managers has the power to bind the Company as provided in this Article VII. No Person dealing with the Company shall have any obligation to inquire into the power or authority of any Person designated by the Board of Managers to act on behalf of the Company.
- 7.8 Indemnification. The Company shall indemnify the Board of Managers and any officers or agents appointed by the Board of Managers for all costs, losses, liabilities and damages paid or incurred in connection with the business of the Company, to the fullest extent provided or allowed by the laws of the State of Delaware.
- 7.9 Resignation and Removal; Replacement. Any member of the Board of Managers may resign upon at least 30 days' prior written notice to the Members. In the event of the resignation or removal of any member of the Board of Managers, the Member who appointed such member of the Board of Managers shall appoint a replacement member of the Board of Managers.
- 7.10 Other Activities. The members of the Board of Managers, in their capacity as such, shall not be required to devote their full time to the management of the Company business, but only so much of their time as the Board of Managers deems necessary or appropriate for the proper management of such business. The members of the Board of Managers, and any of their Affiliates, may engage or possess an interest, independently or with others, in other businesses or ventures of every nature and description, and neither the Company nor any Member shall have any rights in or to such ventures or the income or profits derived therefrom.
- 7.11 **Distributions.** Each Member shall look solely to the assets of the Company for all Distributions and a share of Profits or Losses and shall have no recourse therefor (upon dissolution or otherwise) against the Board of Managers or any of the other Members.
- 7.12 Expenses. The Company shall pay directly or reimburse the Board of Managers for certain expenses of the Company incurred by the Board of Managers in the management of the Company's business. Such expenses may include but are not limited to: (a) costs of borrowed money and taxes applicable to the Company; (b) fees and expenses paid to suppliers, tradespeople, brokers, consultants and other agents; (c) costs of insurance as required in connection with the conduct of the business of the Company; and (d) expenses incurred by the Company for tax return preparation.
- 7.13 Affiliates; Fees. The Board of Managers is specifically authorized to employ, contract and deal with, from time to time, any Member or Affiliate of any Member, and in connection therewith to pay such Person's fees, prices or other compensation; provided that such employment, contracts and dealings are necessary or appropriate for the Company's purposes, and the fees, prices or other compensation paid by the Company therefor is, in the judgment of the Board of Managers, reasonable and typical or competitive with the fees, prices or other compensation customarily paid for similar Property or services.

Nothing herein contained shall be construed as a guaranty by the Board of Managers of the performance by any Affiliate, designee or nominee of its obligations under any contract between any such Affiliate, designee or nominee and the Company.

### ARTICLE VIII CONTRIBUTIONS AND CAPITAL ACCOUNTS

- 8.1 Initial Capital Contributions. Concurrently with the execution and delivery of this Agreement, each Member shall make the Initial Capital Contribution described for that Member on Exhibit A. No Member shall have the right to withdraw or be repaid any Capital Contribution except as provided in this Agreement.
- 8.2 Indemnification. Each Member shall indemnify the other Members and the Company against any and all liabilities of such indemnifying Member which are imputed to the other Members or the Company.
- 8.3 Shares. The Company shall issue Shares to represent each Member's Membership Interest. The total number of Shares that the Company is authorized to issue is one thousand (1,000) Shares. The Company may authorize additional Shares only with the written consent of, and with terms and conditions approved by, the Board of Managers and subject to the provisions of Section 11.5. The initial number of issued and outstanding Shares is set forth on Exhibit A.
- 8.4 Additional Capital Contributions. In the event the Board of Managers determines that the Company does not have sufficient operating revenues or other available funds to pay any amount which the Board of Managers determines to be required for any Company purpose, the Board of Managers shall, if reasonable under the circumstances, attempt to obtain financing in the amount required; provided, however, that the Board of Managers shall not obtain such financing if it would cause a default under any Company obligation. If such financing cannot be obtained within a reasonable time and upon terms and conditions approved by the Board of Managers, the Board of Managers may, pursuant to Section 6.6 above, solicit and accept on behalf of the Company Additional Capital Contributions from the Members. Additional Capital Contributions shall not affect the Sharing Ratios of the Members.
- 8.5 Enforcement of Commitments. In the event any Member (a "Delinquent Member") fails to perform such Delinquent Member's Commitment, the Board of Managers shall give such Delinquent Member a notice of such failure. If the Delinquent Member fails to perform the Commitment (including the payment of any costs associated with the failure and interest at the Default Interest Rate) within ten (10) Business Days of the giving of such notice, the Board of Managers may take such action as it deems appropriate, including but not limited to:
  - (a) Enforcing the Commitment in the court of appropriate jurisdiction in the state in which the Principal Office is located or the state of the Delinquent Member's address as reflected in this Agreement; provided, however, that a Member shall have no personal liability for any Additional Capital Contribution and, in any proceeding to enforce the obligation of a Member to make all or part of any such Additional Capital Contribution, the Board of Managers shall have recourse solely to the Delinquent Member's interest in the Company. Each Member expressly agrees to the jurisdiction of such courts but only for purposes of such enforcement.
    - (b) Selling the Delinquent Member's Shares, including a sale to another Member.
  - (c) Allowing Members, except the Delinquent Member, to make Additional Capital Contributions and adjusting the Sharing Ratios and Shares of the Members in proportion to the new Capital Contribution levels.
    - (d) Reducing the Delinquent Member's Membership Interest and Shares.

- (e) Issuing new Shares to Members who make Additional Capital Contributions in place of the Delinquent Member; provided that such Shares may be entitled to a priority return and such other rights as shall be determined by the Board of Managers.
- 8.6 Capital Account. A separate capital account shall be maintained for each Member throughout the term of the Company in accordance with the rules of section 1.704-1(b)(2)(iv) of the Regulations as in effect from time to time. Except as otherwise provided in section 1.704-1(b)(2)(iv) of the Regulations, each Member's Capital Account shall initially consist of such Member's Initial Capital Contribution, and shall be maintained in accordance with the following provisions:
  - (a) Each Member's Capital Account shall be credited with: (i) such Member's Additional Capital Contributions; and (ii) such Member's share of Profits and items of income and gain that are specially allocated to such Member pursuant to Article IX (other than any income or gain allocated to such Member pursuant to Section 9.3(i) in accordance with section 704(c) of the Code).
  - (b) Each Member's Capital Account shall be debited with: (i) the amount of money distributed to such Member by the Company (including liabilities of such Member assumed by the Company as provided in section 1.704-1(b)(2)(iv)(c) of the Regulations) other than amounts which are in repayment of debt obligations of the Company to such Member; (ii) the Gross Asset Value of Property distributed to such Member (net of liabilities secured by such distributed Property that such Member is considered to assume or take subject to under section 752 of the Code); and (iii) such Member's share of Losses and items of loss and deduction that are specially allocated to such Member pursuant to Article IX (other than any deduction or loss allocated to such Member pursuant to Section 9.3(i) in accordance with section 704(c) of the Code).
  - (c) If the Gross Asset Value of a Company asset differs from its adjusted tax basis, the Members' Capital Accounts shall be adjusted to reflect only allocations to them of depreciation, amortization and gain or loss as computed for book purposes (and not for tax purposes) with respect to such Company asset. In such event, items of book depreciation, amortization and gain or loss shall be calculated in conformity with the rules of section 1.704-1(b)(2)(iv)(g) of the Regulations.
  - (d) All such contributions, allocations and Distributions shall be credited or charged, as the case may be, to the appropriate Capital Accounts of the respective Members to whom they apply as of the time the contributions, allocations or Distributions are made.
  - (e) The Capital Account of a transferee Member shall include the appropriate portion of the Capital Account of the Member from whom the transferee Member's interest was obtained.
  - (f) In determining the amount of any liability, there shall be taken into account section 752 of the Code and any other applicable provisions of the Code and Regulations.

The foregoing provisions and the other provisions of this Agreement relating to the maintenance of Capital Accounts are intended to comply with section 1.704-1(b) of the Regulations, and shall be interpreted and applied in a manner consistent with such Regulations. Consistent with such intention, the value of any Property (other than cash) (i) contributed to the Company by a Member, (ii) distributed to a Member from the Company or (iii) owned by the Company and subject to a revaluation upon the occurrence of certain events shall be the Fair Market Value of such Property (net of liabilities secured by such Property that the Company or such Member, as the case may be, is considered to assume or take

subject to under section 752 of the Code) on the date of contribution, Distribution or revaluation, as applicable.

- 8.7 **No Obligation to Restore Deficit Balance.** No Member shall be required to restore any deficit balance in its Capital Account.
- 8.8 Withdrawal; Successors. A Member shall not be entitled to withdraw any part of its Capital Account or to receive any Distribution from the Company, except as specifically provided in this Agreement. Any Member, including any additional or substitute Member, who shall receive a Membership Interest or whose Membership Interest shall be increased by means of a transfer to it of all or part of the Membership Interest of another Member, shall have a Capital Account with respect to such interest initially equal to the Capital Account with respect to such interest of the Member from whom such interest is acquired.
- 8.9 Interest. Except as otherwise provided in this Agreement, no Member shall be entitled to interest or other return on such Member's Capital Contribution or on any Profits retained by the Company.
- 8.10 Investment of Capital Contributions and Company Cash. The Capital Contributions of the Members and any cash held by the Company from time to time shall be invested, until such time as such funds shall be used for other Company purposes, in demand, money market or time deposits or obligations, securities, investments or other instruments constituting cash equivalents. Such investments shall be made by the Board of Managers for the benefit of the Company.

### 8.11 Repayment of Capital Contribution.

- (a) The members of the Board of Managers shall have no personal liability for the repayment of any Capital Contributions of any Member, and no Member shall have liability for the repayment of any Capital Contributions of any other Member. The repayment of any Capital Contribution shall be made only to the extent of available Company assets in accordance with the terms of this Agreement.
- (b) Except as otherwise provided in this Agreement, no Member shall have priority over any other Member as to the return of its Capital Contribution or as to Distributions of cash made by the Company.
- (c) Except as otherwise provided in this Agreement, a Member shall not be entitled to (i) demand or receive Property other than cash in return for its Capital Contribution or (ii) receive any funds or Property of the Company.

### ARTICLE IX ALLOCATIONS AND DISTRIBUTIONS

- 9.1 Profits and Losses. Profits and Losses, and each item of Company income, gain, loss, deduction, credit and tax preference with respect thereto, for each Fiscal Year (or shorter period in respect of which such items are to be allocated) shall be allocated among the Members as provided in this Article IX.
- 9.2 Allocation of Profits and Losses. All items of Profits and Losses for any Fiscal Year shall be allocated to the Members in proportion to their respective Sharing Ratios.

16

### 9.3 **Special Allocations.** The following special allocations shall be made:

- (a) Loss Limitation. The Losses allocated pursuant to Section 9.2 shall not exceed the maximum amount of Losses that may be allocated to a Member without causing such Member to have an Adjusted Capital Account Deficit at the end of the Fiscal Year. If some, but not all of the Members would have adjusted Capital Account Deficits as a consequence of the allocations of Losses pursuant to Section 9.2, the limitation in this Section 9.3(a) shall be applied by allocating Losses pursuant to Section 9.3(a) only to those Members (allocated pro rata if more than one) who would not have an Adjusted Capital Account Deficit as a consequence of receiving such an allocation. All Losses in excess of the limitations set forth in this Section 9.3(a) shall be allocated among the Members in proportion to their respective Sharing Ratios.
- (b) Minimum Gain Chargeback. Except as otherwise provided in section 1.704-2(f) of the Regulations, notwithstanding any other provision of this Article IX, if there is a net decrease in Partnership Minimum Gain during any Fiscal Year, each Member shall be specially allocated items of Company income and gain for such Fiscal Year (and, if necessary, subsequent Fiscal Years) in an amount equal to such Member's share of the net decrease in Partnership Minimum Gain, determined in accordance with section 1.704-2(g) of the Regulations. Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Member pursuant thereto. The items to be so allocated shall be determined in accordance with sections 1.704-2(f)(6) and 1.704-2(j)(2) of the Regulations. This Section 9.3(b) is intended to comply with the minimum gain chargeback requirement in section 1.704-2(f) of the Regulations and shall be interpreted consistently therewith.
- (c) Partner Minimum Gain Chargeback. Except as otherwise provided in section 1.704-2(i)(4) of the Regulations, notwithstanding any other provision of this Article IX, if there is a net decrease in Partner Nonrecourse Debt Minimum Gain attributable to a Partner Nonrecourse Debt during any Fiscal Year, each Member who has a share of the Partner Nonrecourse Debt Minimum Gain as of the beginning of the Fiscal Year attributable to such Partner Nonrecourse Debt, determined in accordance with section 1.704-2(i)(5) of the Regulations, shall be specially allocated items of Company income and gain for such Fiscal Year (and, if necessary, subsequent Fiscal Years) in an amount equal to such Member's share of the net decrease in Partner Nonrecourse Debt Minimum Gain attributable to such Partner Nonrecourse Debt, determined in accordance with section 1.704-2(i)(4) of the Regulations. Allocations pursuant to the previous sentence shall be made in proportion to the respective amounts required to be allocated to each Member pursuant thereto. The items to be so allocated shall be determined in accordance with sections 1.704-2(i)(4) and 1.704-2(j)(2) of the Regulations. This Section 9.3(c) is intended to comply with the minimum gain chargeback requirement in section 1.704-2(i)(4) of the Regulations and shall be interpreted consistently therewith.
- (d) Qualified Income Offset. In the event any Member unexpectedly receives any adjustments, allocations or Distributions described in section 1.704-1(b)(2)(ii)(d)(4), section 1.704-1(b)(2)(ii)(d)(5) or section 1.704-1(b)(2)(ii)(d)(6) of the Regulations which increase a Member's Adjusted Capital Account Deficit, items of Company income and gain shall be specially allocated to the Member in an amount and manner sufficient to eliminate, to the extent required by the Regulations, the Adjusted Capital Account Deficit of the Member as quickly as possible, provided that an allocation pursuant to this Section 9.3(d) shall be made only if and to the extent that the Member would have an Adjusted Capital Account Deficit after all other allocations provided for in this Article IX have been tentatively made as if this Section 9.3(d) were not in this Agreement.

- (e) Gross Income Allocation. In the event any Member has an Adjusted Capital Account Deficit, such Member shall be specially allocated items of Company income and gain in the amount of such excess as quickly as possible, provided that an allocation pursuant to this Section 9.3(e) shall be made only if and to the extent that such Member would have an Adjusted Capital Account Deficit after all other allocations provided for in this Article IX have been made as if Section 9.3(d) and this Section 9.3(e) were not in this Agreement.
- (f) <u>Nonrecourse Deductions</u>. Nonrecourse Deductions for any Fiscal Year shall be specially allocated among the Members in proportion to their Sharing Ratios.
- (g) <u>Partner Nonrecourse Deductions</u>. Any Partner Nonrecourse Deductions for any Fiscal Year shall be specially allocated to the Member who bears the economic risk of loss with respect to the Partner Nonrecourse Debt to which such Partner Nonrecourse Deductions are attributable in accordance with section 1.704-2(i)(1) of the Regulations.
- (h) Certain Book-ups. To the extent an adjustment to (i) the adjusted tax basis of any Company asset pursuant to section 734(b) or section 743(b) of the Code is required to be taken into account in determining Capital Accounts or (ii) pursuant to section 1.704-1(b)(2)(iv)(f) of the Regulations, the Gross Asset Value of any Company asset is permitted to be taken into account in determining Capital Accounts, the amount of such adjustment to the Capital Accounts shall be treated, as provided in section 1.704-1(b)(2)(iv)(m) or section 1.704-1(b)(2)(iv)(g) of the Regulations, respectively, as an item of Profit (if the adjustment increases such basis or Gross Asset Value of the asset) or Loss (if the adjustment decreases such basis or Gross Asset Value), and such Profit or Loss shall be specially allocated to the Members in a manner consistent with the manner in which their Capital Accounts are required to be adjusted pursuant to such section of the Regulations.

### (i) Mandatory Allocations under Section 704(c) of the Code.

- (i) In the event section 704(c) of the Code or the principles of section 704(c) of the Code applicable under section 1.704-1(b)(2)(iv) of the Regulations require allocations of income, gain, deduction or loss in a manner different than that set forth above, the provisions of section 704(c) of the Code and the Regulations thereunder shall control such allocations among the Members. Any item of Company income, gain, loss and deduction with respect to any Property (other than cash) that has been contributed by a Member to the capital of the Company and which is required or permitted to be allocated to such Member for income tax purposes under section 704(c) of the Code so as to take into account the variation between the tax basis of such Property and its Fair Market Value at the time of its contribution shall be allocated solely for income tax purposes in the manner so required or permitted under section 704(c) of the Code using the "traditional method" described in section 1.704-3(b) of the Regulations; provided, however, that any other method allowable under applicable Regulations may be used for any contribution of Property if the contributing Member and the Board of Managers agree as to the use of such other method,
- (ii) In the event the Gross Asset Value of any Company asset is adjusted pursuant to Section 9.3(h) in accordance with section 1.704-1(b)(2)(iv)(f) of the Regulations, subsequent allocations of income, gain, loss and deduction with respect to such asset shall take account of any variation between the adjusted basis of such asset for federal income tax purposes and its Gross Asset Value in a manner consistent with Section 9.3(i)(i).

- (iii) Except as provided in Sections 9.4(h)(i) and (ii), for United States federal, state and local income tax purposes, the income, gains, losses and deductions of the Company shall, for each taxable period, be allocated among the Members in the same manner and in the same proportion that such items have been allocated among the Members' respective Capital Accounts.
- 9.4 Curative Allocations. The Regulatory Allocations are intended to comply with certain requirements of the Regulations. It is the intent of the Members that, to the extent possible, all Regulatory Allocations shall be offset either with other Regulatory Allocations or with special allocations of other items of Company income, gain, loss or deduction. Therefore, notwithstanding any other provision of this Article IX (other than the Regulatory Allocations), the Board of Managers shall make such offsetting special allocations of Company income, gain, loss or deduction in whatever manner it determines appropriate so that, after such offsetting allocations are made, each Member's Capital Account balance is, to the extent possible, equal to the Capital Account balance such Member would have had if the Regulatory Allocations were not part of this Agreement and all Company items were allocated pursuant to Section 9.2. In exercising its discretion under this Section 9, the Board of Managers shall take into account future Regulatory Allocations under Sections 9.3(b) and 9.3(c) that, although not yet made, are likely to offset other Regulatory Allocations previously made under Sections 9.3(f) and 9.3(g).
- 9.5 Section 754 Election. Upon a transfer by a Member of an interest in the Company, which transfer is permitted by the terms of this Agreement, or upon the death of a Member or the Distribution of any Company Property to one or more Members, the Board of Managers, upon the request of one or more of the transferees or distributees, shall cause the Company to file an election on behalf of the Company to cause the basis of the Company's Property to be adjusted for federal income tax purposes in the manner prescribed in section 734 or 743 of the Code, as the case may be. The cost of preparing such election and any additional accounting expenses of the Company occasioned by such election shall be borne by such transferees or distributees.

### 9.6 Other Allocation Rules.

- (a) For purposes of determining the Profits, Losses or any other item allocable to any period (including allocations to take into account any changes in any Member's Sharing Ratio during a Fiscal Year and any transfer of any interest in the Company), Profits, Losses and any such other item shall be determined on a daily, monthly or other basis, as determined by the Board of Managers using any permissible method under section 706 of the Code and the Regulations thereunder.
- (b) Notwithstanding the above, allocations made pursuant to this Article IX shall be made annually within one hundred and five (105) days after the end of the Fiscal Year, but in no event may such allocations be made later than required by the Code.
- (c) The Members are aware of the income tax consequences of the allocations made by this Article IX and hereby agree to be bound by the provisions of this Article IX in reporting their shares of Company income and loss for income tax purposes.
- (d) Solely for purposes of determining a Member's proportionate share of the "excess nonrecourse liabilities" of the Company within the meaning of section 1.752-3(a)(3) of the Regulations, the Members' interests in Company Profits are in proportion to their Sharing Ratios.

- (e) To the extent permitted by section 1.704-2(h)(3) of the Regulations, the Board of Managers shall endeavor to treat Distributions made from the proceeds of a Nonrecourse Liability or a Partner Nonrecourse Debt as Distributions that are not allocable to an increase in minimum gain to the extent that such Distributions would not cause or increase an Adjusted Capital Account Deficit for any Member.
- (f) Except as otherwise provided in this Article IX, an allocation of Company Profits or Losses to a Member shall be treated as an allocation to such Member of the same share of each item of income, gain, loss and deduction taken into account in computing such Profits or Losses.
- (g) For purposes of determining the character (as ordinary income or capital gain) of any Profits allocated to the Members pursuant to this Article IX, such portion of Profits that is treated as ordinary income attributable to the recapture of depreciation shall, to the extent possible, be allocated among the Members in the proportion which (i) the amount of depreciation previously allocated to each Member bears to (ii) the total of such depreciation allocated to all Members. This Section 9.6(g) shall not alter the amount of allocations among the Members pursuant to this Article IX, but merely the character of income so allocated.
- (h) Except for arrangements expressly described in this Agreement, no Member shall enter into (or permit any Person related to the Member to enter into) any arrangement with respect to any liability of the Company that would result in such Member (or a person related to such Member under section 1.752-4(b) of the Regulations) bearing the economic risk of loss (within the meaning of section 1.752-2 of the Regulations) with respect to such liability unless such arrangement has been approved by all Members. To the extent a Member is permitted to guarantee the repayment of any Company indebtedness under this Agreement, each of the other Members shall be afforded the opportunity to guarantee such Member's pro rata share of such indebtedness, determined in accordance with the Members' respective Sharing Ratios.

#### 9.7 Distribution of Net Cash Flow.

- (a) Amounts and Timing. Subject to the provisions of Section 13.3 and applicable law, Net Cash Flow for each Fiscal Year of the Company, shall be distributed to the Members in proportion to their respective Sharing Ratios within 30 days of the completion of the Annual Independent Audit, or at such earlier time or times as may be designated by the Board of Managers.
- (b) Amounts Withheld. The Board of Managers is authorized to withhold from Distributions, or with respect to allocations, to the Members and to pay over to any federal, state or local government any amounts required to be so withheld pursuant to the Code or any provisions of any other federal, state or local tax law. Any amounts required to be withheld pursuant to the Code or any provision of any state or local tax law with respect to any payment, distribution or allocation to the Company or the Members shall be treated as an amount distributed to the Member with respect to which such amount was withheld pursuant to this Section 9.7 for all purposes under this Agreement.
- (c) <u>Draws for Payment of Estimated Taxes</u>. The Company shall pay to each Member an annual draw, not to exceed the amount reasonably necessary to provide for payment by the Members of any federal, state and local estimated taxes with respect to Profits allocated to the Members pursuant to this Article IX, and each such draw, if any, shall be treated as a loan from the Company to each Member receiving such draw and shall be deemed repaid by reducing

the amount of each subsequent Distribution to the Member receiving such draw pursuant to this Section 9.7 by the lesser of (i) the entire amount otherwise distributable to the Member receiving such draw and (ii) the entire amount of any unrepaid draws pursuant to this Section 9.7(c).

### 9.8 Allocation of Gain or Loss upon Winding Up.

- (a) Gain or Loss Realized on Sale of Company Property. Upon the winding up of the Company, as provided in Article XIII, net gain or loss realized on the sale or sales of Company Property (each such sale a "Winding Up Sale") shall be allocated among the Members in accordance with Section 9.2.
- (b) <u>Distributions in Kind</u>. In the event that, upon a winding up of the Company, pursuant to Section 13.3, either (i) any Company Property is required to be distributed in kind, or (ii) the Board of Managers elects to distribute any Company Property in kind, the book value of such Property shall be adjusted to its Fair Market Value as of the date of such Distribution, and the amount of such adjustments shall be allocated to the Members' Capital Accounts in the manner provided in Section 9.8(a) as though such Property had been sold at its Fair Market Value and gain or loss had been realized.

### ARTICLE X TAXES

Tax Matters Partner. Coffee Holding shall be the Tax Matters Partner of the Company pursuant to section 6231(a)(7) of the Code. If Coffee Holding shall cease to act as the Tax Matters Partner for any reason, the Members shall select another Member (subject to such Member's approval) to be the Tax Matters Partner. The Tax Matters Partner shall receive no additional compensation from the Company for its services in that capacity, but all expenses incurred by the Tax Matters Partner in such capacity shall be borne by the Company. The Tax Matters Partner is authorized to employ such accountants, attorneys and agents as it, in its sole discretion, determines is necessary to or useful in the performance of its duties. The Tax Matters Partner is authorized to represent the Company before the Internal Revenue Service and any other governmental agency with jurisdiction, and to sign such consents and to enter into settlements and other agreements with such agencies as the Tax Matters Partner or its duly authorized officer deems necessary or advisable. Each Member shall give prompt notice to each other Member of any and all notices it receives from the Internal Revenue Service concerning the Company, including any notice of a 30 day appeal letter and any notice of deficiency in tax concerning the Company's federal income tax returns. The Tax Matters Partner shall give each Member periodic status reports regarding any negotiations between the Internal Revenue Service and the Company. The Tax Matters Partner shall serve in a similar capacity with respect to any similar tax related or other election provided by state or local laws.

### ARTICLE XI TRANSFER OF MEMBERSHIP INTEREST

11.1 Compliance with Securities Laws and this Agreement. No Shares have been registered under the Securities Act or under any applicable state securities laws. A Member may not transfer (a transfer, for purposes of this Agreement, shall be deemed to include, but not be limited to, any sale, transfer, assignment, pledge, creation of a security interest or other disposition) any Shares, except upon compliance with the applicable federal and state securities laws and this Agreement. The Board of Managers shall have no obligation to register any Shares under the Securities Act or under any applicable state securities laws, or to make any exemption therefrom available to any Member except to the extent provided in any Registration Rights Agreement entered into by the Company with respect thereto.

21

- 11.2 Transfer of Shares; Admission of Substitute Member. Except for those transfers permitted by this Article XI, a Member may transfer Shares and give the transferee the right to become a Member only after the following terms and conditions have been satisfied:
  - (a) The transferee shall also be the transferee of all or part of the transferor's Economic Interest or shall be the owner of an Economic Interest;
  - (b) The Members holding all of the Shares that are not the subject of the transfer shall have consented in writing to the transfer of such Shares and the admission of the Substitute Member;
  - (c) The transferor and the transferee shall have complied with such other requirements as the non-transferring Members may reasonably impose, including the conditions that the transferee:
    - (i) Adopt and approve in writing all the terms and provisions of this Agreement then in effect; and
    - (ii) Pay such fees as the Board of Managers may reasonably require to pay the costs of the Company in effecting such substitution; and
    - (d) The Board of Managers shall have consented to the transfer.
- 11.3 **Dispositions Not in Compliance with this Article Void.** Any attempted Disposition of a Membership Interest, or any part thereof, not in compliance with this Article shall be void when made and ineffectual and shall not bind the Company.
- 11.4 Transfer to Affiliate of a Member. Notwithstanding anything herein to the contrary, any Member may transfer its Shares to any Affiliate or direct or indirect subsidiary thereof that is controlled by or under common control with it, and any such transferee shall have the right to become a Substitute Member, subject to compliance with Section 11.1.

# 11.5 Future Sales and Preemptive Rights.

(a) The Company will not offer or sell any New Securities (an "Equity Offering") unless it first offers to each Member the right to purchase a portion of such New Securities which is equal to the total number of New Securities to be sold in the Equity Offering multiplied by a fraction, the numerator of which is the number of Shares owned by such Member on a fully diluted basis, and the denominator of which is the total number of Shares of the Company then outstanding on a fully diluted basis (an "Offering Allocation"). The Company shall immediately notify each Member of the terms or the proposed terms of an Equity Offering (the "Company's Notice"). The Company's Notice shall include the Offering Allocation for each Member. If the Member wishes to purchase New Securities pursuant to the Company's Notice, such Member shall notify the Company in writing within thirty (30) days after receipt of the Company's Notice stating how many of such New Securities such Member desires to purchase. Each Member shall be entitled to purchase up to that number of New Securities equal to such Member's respective Offering Allocation. To the extent the Members do not elect to purchase any or all of such New Securities, the Company shall have up to ninety (90) days to complete the Equity Offering upon the same terms specified in the Company's Notice. If the Company later changes the terms of the Equity Offering in any material respect, the Company shall first re-offer such New Securities to the Members pursuant to the procedure set forth above. Any New Securities which are offered or sold (or issued) by the Company in the Equity Offering shall, unless otherwise consented to in writing by each Member, be sold to any purchaser only for cash.

(b) The closing of the purchase of New Securities by the Members pursuant to this Section 11.5 shall occur at the Principal Office of the Company concurrently with the closing of the Equity Offering, or at such other time as the Company and the Members may mutually determine.

# ARTICLE XII DISSOCIATION OF A MEMBER

- 12.1 **Dissociation.** A Person shall cease to be a Member upon the happening of any of the following events:
  - (a) The resignation or withdrawal of the Member;
  - (b) The Member becoming a Bankrupt Member;
  - (c) The Member breaching this Agreement and not curing such breach within sixty (60) days of a non-breaching Member providing written notice of such breach pursuant to Section 13.1(c) below;
  - (d) In the case of a Member who is a natural person, the death of the Member or the entry of an order by a court of competent jurisdiction adjudicating the Member incompetent to manage the Member's personal estate;
  - (e) In the case of a Member that is a trust or who is acting as a Member by virtue of being a trustee of a trust, the termination of the trust (but not merely the substitution of a new trustee);
  - (f) In the case of a Member that is a separate Organization other than a corporation, the dissolution and commencement of winding up of the separate Organization;
  - (g) In the case of a Member that is a corporation, the filing of a certificate of dissolution, or its equivalent, for the corporation or the revocation of its charter; or
  - (h) In the case of a Member that is an estate, the distribution by the fiduciary of the estate's entire interest in the Company.
- Rights of Dissociating Member. In the event any Member dissociates prior to the expiration of the term of this Agreement, such Member shall be entitled to participate in the dissolution and winding up of the Company pursuant to Article XIII, to the same extent as any other Member except that, if such Dissociation results from a withdrawal of the Member in violation of this Agreement, any Distributions to which such Member would have been entitled shall be reduced by that portion of the damages, if any, sustained by the Company as a result of the Dissolution Event and winding up that is chargeable to the Capital Accounts of the other Members.

23

# ARTICLE XIII DISSOLUTION AND WINDING UP

- 13.1 **Dissolution.** The Company shall be dissolved without further action by the Members and its affairs wound up upon the first to occur of any of the following events (each of which shall constitute a Dissolution Event):
  - (a) The unanimous written consent of the Members;
  - (b) A Member materially breaches this Agreement and a non-breaching Member provides sixty (60) days written notice to the other Members of such breach, during which sixty (60) day period, the breaching Member shall have the opportunity to cure such breach; provided, however, that, the other Members shall have the opportunity to elect to continue the Company without the breaching Member, in which case the breaching Member shall be dissociated pursuant to Article XII above;
  - (c) A Member ceases to be a Member pursuant to Article XII above; provided, however, that, the other Members shall have the opportunity to elect to continue the Company without such Member;
  - (d) A Member becomes a Bankrupt Member; provided, however, that, the other Members shall have the opportunity to elect to continue the Company without such Member;
  - (e) A material breach of any agreement between the Company and a Member or between Members; or
    - (f) A Member elects to dissolve the Company pursuant to Section 14.12 below.
- 13.2 **Effect of Dissolution.** Upon dissolution, the Company shall not be terminated and shall continue until the winding up of the affairs of the Company is completed and a certificate of dissolution has been issued by the Secretary of State of the State of Delaware.
- 13.3 **Distribution of Assets on Dissolution.** Upon the winding up of the Company, the Board of Managers shall take full account of the assets and liabilities of the Company, and shall apply and distribute the assets and any proceeds therefrom in the following order:
  - (a) First, to the payment of the debts and liabilities of the Company to creditors, including Members and affiliates of Members who are creditors, to the extent permitted by law, in satisfaction of such debts and liabilities, and to the payment of necessary expenses of liquidation;
  - (b) Second, to the setting up of any reserves which the Board of Managers may deem necessary or appropriate for any anticipated obligations or contingencies of the Company arising out of or in connection with the operation or business of the Company. Such reserves may be paid over by the Board of Managers to an escrow agent or trustee selected by the Board of Managers to be disbursed by such escrow agent or trustee in payment of any of the aforementioned obligations or contingencies and, if any balance remains at the expiration of such period as the Board of Managers shall deem advisable, shall be distributed by such escrow agent or trustee in the manner hereinafter provided; and

(c) Then, to the Members in accordance with positive Capital Account balances taking into account all Capital Account adjustments for the Company's taxable year in which the liquidation occurs. Liquidation proceeds shall be paid within sixty (60) days of the end of the Company's taxable year in which the liquidation occurs. Such Distributions shall be in cash or Property (which need not be distributed proportionately) or partly in both, as determined by the Board of Managers.

To the extent possible after making all distributions under Sections 13.3(a) and (b) (and without violating Section 13.3(c)), the assets contributed to the Company by each Member shall be distributed to such contributing Member. In the event that the Company is unable to comply with the obligations set forth in Sections 13.3(a) and (b) above without liquidating such assets, the Board of Managers shall determine which assets to liquidate to comply with such obligations. In such case, the remainder of the assets shall be distributed to the contributing Member of such assets.

13.4 Winding Up and Certificate of Dissolution. The winding up of the Company shall be completed when all debts, liabilities and obligations of the Company have been paid and discharged or reasonably adequate provision therefor has been made, and all of the remaining Property of the Company has been distributed to the Members. Upon the completion of the winding up of the Company, a certificate of dissolution shall be delivered to the Secretary of State of the State of Delaware for filing. The certificate of dissolution shall set forth the information required by the Act.

# ARTICLE XIV MISCELLANEOUS

- 14.1 Notices. Notices to the Board of Managers shall be sent to the Principal Office of the Company with a copy to Matthew Dyckman, Esq., Thacher Proffitt & Wood LLP, 1700 Pennsylvania Avenue, NW, Suite 800, Washington, DC 20006. Notices to the other Members shall be sent to their addresses set forth on Exhibit A. Any Member may require notices to be sent to a different address by giving notice to the other Members in accordance with this Section 14.1. Any notice or other communication required or permitted hereunder shall be in writing, and shall be deemed to have been given with receipt confirmed if and when delivered personally, or when sent by a recognized overnight delivery service, delivered by courier, or sent by facsimile, to such Members at such address.
- 14.2 Headings. All Article and Section headings in this Agreement are for convenience of reference only and are not intended to qualify the meaning of any Article or Section.
- 14.3 Entire Agreement. This Agreement constitutes the entire agreement among the parties and supersedes any prior agreement or understanding among them respecting the subject matter of this Agreement except as specifically provided herein.
- 14.4 **Binding Agreement.** This Agreement shall be binding upon, and inure to the benefit of, the parties hereto, their successors, heirs, legatees, devisees, assigns, legal representatives, executors and administrators, except as otherwise provided herein.
- 14.5 Saving Clause. If any provision of this Agreement, or the application of such provision to any Person or circumstance, shall be held invalid, the remainder of this Agreement, or the application of such provision to Persons or circumstances other than those as to which it is held invalid, shall not be

affected thereby. If the operation of any provision of this Agreement would contravene the provisions of the Act, such provision shall be void and ineffectual.

- 14.6 Counterparts. This Agreement may be executed in several counterparts, and all so executed shall constitute one agreement, binding on all the parties hereto, even though all parties are not signatories to the original or the same counterpart. Any counterpart of either this Agreement or the Certificate of Formation shall for all purposes be deemed a fully executed instrument. For purposes of this Agreement, facsimile as well as scanned and emailed signatures shall be deemed to be original signatures.
- 14.7 Governing Law; Jurisdiction; Venue. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware, without regard to the conflicts of laws principles thereof. To the fullest extent permitted by law, the parties hereto hereby (i) submit to the jurisdiction of the state and federal courts located in the State of Delaware for purposes of any legal action or proceeding brought under or in connection with this Agreement, (ii) agree that exclusive venue of any such action or proceeding may be laid in the State of Delaware and (iii) waive any claim that the same is an inconvenient forum.
- 14.8 No Partnership Intended for Nontax Purposes. The Members have formed the Company under the Act, and expressly do not intend hereby to form a partnership under either the Delaware Uniform Partnership Act or the Delaware Uniform Limited Partnership Act. The Members do not intend to be partners one to another or partners as to any third party. To the extent any Member, by word or action, represents to another person that any Member is a partner or that the Company is a partnership, the Member making such wrongful representation shall be liable to any other Members who incur personal liability by reason of such wrongful representation.
- 14.9 No Rights of Creditors and Third Parties. This Agreement is entered into among the Company and the Members for the exclusive benefit of the Company, its Members and their permitted successors and assignees. This Agreement is expressly not intended for the benefit of any creditor of the Company or any other Person. Except and only to the extent provided by applicable statute, no such creditor or any third party shall have any rights under this Agreement or any agreement between the Company and any Member with respect to any Capital Contribution or otherwise.

#### 14.10 Customers/Non-compete.

- (a) In the event that the Company is dissolved, each Member shall retain the exclusive right to conduct business with the customers such Member introduced to the Company and the Members agree to divide the Company's new customers evenly among the Members.
- 14.11 Confidentiality. It is recognized that during their work on behalf of the Company, the Members may have occasion to conceive, create, develop, review, or receive information that is considered by the Company or another Member to be confidential or proprietary, including without limitation information relating to the business of the Company, including inventions, patent, trademark and copyright applications, improvements, know-how, specifications, drawings, cost data, process flow diagrams, customer and supplier lists, bills, ideas, and/or any other written material referring to the same ("Confidential Information"). Both during the term of this Agreement and thereafter, the Members agree to maintain the confidence of the Confidential Information unless or until such Confidential Information has been made public by an act or omission of a Member other than itself.

### 14.12 Dispute Resolution.

- (a) Whenever the Members shall have any dispute among themselves relating to the interpretation, construction or implementation of this Agreement, the Members shall resolve such dispute as follows:
  - (i) First, each Member involved in such dispute shall use its good faith efforts to negotiate a resolution thereof by engaging in discussions with the other Members so involved at reasonable times and places, by telephone or otherwise, during the sixty (60) day period following notice by a Member to each of the other Members of its belief that there is a dispute which requires resolution in such manner;
  - (ii) Second, if the Members are unable to resolve such dispute through good faith negotiations during the sixty (60) day period provided in Section 14.11(a)(i), the Members shall refer the dispute to mediation in accordance with the Judicial Arbitration and Mediation Services, Inc. (J·A·M·S) (www.jamsadr.com) mediation procedures; and
  - (iii) Third, if the Members are unable to resolve such dispute through mediation, any Member involved in such dispute may bring an action or proceeding in any court having jurisdiction thereof pursuant to Section 14.7; provided that each Member waives its right to trial by jury and its right to consequential, special and/or punitive damages.
- (b) Whenever the Members shall be deadlocked or shall otherwise be in dispute with respect to the relations among the Members or between the Members and the Company or any other matter related thereto, the Members shall resolve such dispute as follows:
  - (i) First, each Member involved in such dispute shall use its good faith efforts to negotiate a resolution thereof by engaging in discussions with the other Members so involved at reasonable times and places, by telephone or otherwise, during the sixty (60) day period following notice by a Member to each of the other Members of its belief that there is a dispute which requires resolution in such manner;
  - (ii) Second, if the Members are unable to resolve such dispute through good faith negotiations during the sixty (60) day period provided in Section 14.12(b)(i), the Members shall refer the dispute to mediation in accordance with the Judicial Arbitration and Mediation Services, Inc. (J·A·M·S) (www.jamsadr.com) mediation procedures; and
  - (iii) Third, if the Members are unable to resolve such dispute through mediation, any Member involved in such dispute may elect to dissolve the Company, in which case the Company shall dissolve in accordance with Article XIII above.
- 14.13 General Interpretive Principles. For purposes of this Agreement, except as otherwise expressly provided or unless the context otherwise requires:
  - (a) The terms defined in this Agreement include the plural as well as the singular;
  - (b) Accounting terms not otherwise defined herein have the meanings given to them in accordance with generally accepted accounting principles in the United States;

- (c) References herein to "Sections," "paragraphs" and other subdivisions without reference to a document are to designated Sections, paragraphs and other subdivisions of this Agreement;
- (d) A reference to a paragraph without further reference to a Section is a reference to such paragraph as contained in the same Section in which the reference appears, and this rule shall also apply to other sub-divisions;
- (e) The words "herein," "hereof," "hereunder" and other words of similar import refer to this Agreement as a whole and not to any particular provision; and
- (f) The term "include" or "including" shall mean without limitation by reason of enumeration.

IN WITNESS WHEREOF, the parties hereto have executed this Agreement as of the Effective Date.

# THE COMPANY:

Generations Coffee Company, LLC

By: Name: NAMEW GONDAN

Title: parmer

MEMBERS:

Coffee Holding Co, Inc

By: Name: ANNEW GORDON Title: PRENOENT/CEO

Cardso Colfee

Name: Dominic CARUSO

Title: CFO

[IPW: NYUEGIAL:481186 6] 19499-00010 04/06/2006 05:29 PM

#### **EXHIBIT A**

Name and Address of Member	Initial Membership Interest (as represented by number of Shares)	Initial Sharing Ratio
Coffee Holding Co., Inc.  4401 First Avenue Brooklyn, NY 11232 Attn: Andrew Gordon, President and Chief Executive Officer  Phone: (718) 832-0800 Fax: (718) 832-0892 Email: agordon@coffeeholding.com	600*	60%*
Caruso's Coffee  10300 Brecksville, Road Brecksville, OH 44141 Attn: Dominic Caruso  Phone: (800) 562-8575 Fax: [Fax Number] Email: dominic@carusoscoffee.com	400*	40%*
TOTAL	1,000	100%

<sup>\*</sup>In the event that Caruso's Coffee makes additional capital contributions so that the fair market value of its Capital Contributions are equal to the fair market value of Coffee Holding's Capital Contributions, which it may do at any time in its sole discretion, the Initial Membership Interests and Initial Sharing Ratios of the Members shall be adjusted so that each Member shall own 500 shares and have an Initial Sharing Ratio of 50%; provided, however, that any such event shall be documented in an amendment to this Agreement.

[TPW: NYLEGAL:481186.7] 19499-00010 04/07/2006 02:17 PM

# EXHIBIT B

The initial Board of Managers of the Company shall consist of the following persons:

- 1. Dominic Caruso
- 2. Andrew Gordon

[TPW: NYLEGAL:481186.7] 19499-00010 04/07/2006 02:17 PM

# EXHIBIT C

# Capital Contributions of Coffee Holding Co, Inc.

# <u>Item</u>

1. ICA CSV40 Package Machine for Ground and Whole Bean Coffee and related parts.

# **EXHIBIT D**

# Capital Contributions of Caruso's Coffee

# <u>Item</u>

- 1. Use of equipment owned by Caruso's Coffee
- 2. Use of plant and warehouse space owned or leased by Caruso's Coffee

[TPW: NYLEGAL:481186.7] 19499-00010 04/07/2006 02:17 PM

# EXHIBIT 2

# 900487913 03/04/2019

# 900487913 03/04/2019

# TRADEMARK ASSIGNMENT COVER SHEET

Electronic Version v1.1 Stylesheet Version v1.2 ETAS ID: TM512510

SUBMISSION TYPE:	NEW ASSIGNMENT		
NATURE OF CONVEYANCE:	ASSIGNMENT OF THE ENTIRE INTEREST AND THE G	GOODWILL	

# **CONVEYING PARTY DATA**

Name	Formerly	Execution Date	Entity Type
Steep & Brew, Inc.	_	04/24/2018	Corporation: WISCONSIN

# **RECEIVING PARTY DATA**

Name:	Generations Coffee Company, LLC 850 New Burton Rd, Suite 201 ,		
Street Address:			
City:	Dover		
State/Country:	DELAWARE		
Postal Code:	19904		
Entity Type: Limited Liability Company: DELAWARE			

# **PROPERTY NUMBERS Total: 18**

Property Type	Number	Word Mark		
Registration Number:	1553130	STEEP & BREW		
Registration Number:	2078262	WORLD CUP		
Registration Number:	4522990	KENDRA		
Registration Number:	4297249	CAFÉ FAIR		
Registration Number:	3816076	COFFEE DONE RIGHT		
Registration Number:	1971360	WHOLE BEAN STEEP & BREW COFFEES		
Registration Number:	1968491	WHOLE BEAN STEEP & BREW COFFEES		
Registration Number:	2076657	WORLD CUP		
Registration Number:	2706420	WORLD CUP - HAWAII MEXICO GUATEMALA EL S		
Registration Number:	4502625	K2 COFFEE		
Registration Number:	4086431	CLUB FUND ZONE		
Registration Number:	3812456	COFFEE DONE RIGHT		
Registration Number:	2774681	TOASTED NUT		
Registration Number:	1955449	TOASTED NUT		
Registration Number:	1960051	WE KNOW BEANS ABOUT COFFEE		
Registration Number:	1892293	WE KNOW BEANS ABOUT COFFEE		
Registration Number:	2082016	COFFEE CRAFTSMEN		
Registration Number:	2051356	COFFEE CRAFTSMEN		

TRADEMARK
REEL: 006589 FRAME: 0346
TRADEMARK

REEL: 008317 FRAME: 0600

OP \$465.00 1553130

**CORRESPONDENCE DATA** 

**Fax Number:** 6142322410

Correspondence will be sent to the e-mail address first; if that is unsuccessful, it will be sent

using a fax number, if provided; if that is unsuccessful, it will be sent via US Mail.

**Phone:** 6142322628

Email: aperko@reminger.com

Correspondent Name: Acacia Perko

Address Line 1: 200 Civic Center Drive

Address Line 2: Ste 800

Address Line 4: Columbus, OHIO 43215

NAME OF SUBMITTER:	Acacia Perko, Attorney of Record
SIGNATURE:	/Acacia Perko/
DATE SIGNED:	03/04/2019

# **Total Attachments: 6**

source=SteepBrew_Trademark_Assignment_Agreement#page1.tif
source=SteepBrew_Trademark_Assignment_Agreement#page2.tif
source=SteepBrew_Trademark_Assignment_Agreement#page3.tif
source=SteepBrew_Trademark_Assignment_Agreement#page4.tif
source=SteepBrew_Trademark_Assignment_Agreement#page5.tif
source=SteepBrew_Trademark_Assignment_Agreement#page6.tif

# **ASSIGNMENT OF TRADEMARKS**

This **ASSIGNMENT OF TRADEMARKS** (this "<u>Agreement</u>"), dated as of April 24, 2018 (the "<u>Effective Date</u>"), is entered into by and between Generations Coffee Company, LLC (the "<u>Assignee</u>") and Steep & Brew, Inc., a Wisconsin corporation (the "<u>Assignor</u>"). Capitalized terms used but not defined herein shall have the meanings ascribed to such terms in the Purchase Agreement (defined below).

#### **RECITALS**

**WHEREAS**, the Assignor, the Assignee and certain other parties thereto have entered into an Asset Purchase Agreement, dated as of the date hereof (as it may be amended, amended and restated, or otherwise modified from time to time, the "<u>Purchase Agreement</u>"); and

WHEREAS, pursuant to the Purchase Agreement, the Assignor has agreed to transfer and assign to the Assignee all of the Assignor's right, title and interest in, to and under the Assignor's registered and unregistered marks and/or trade names existing as of the Effective Date, including, without limitation, those set forth on the attached <u>Schedule A</u> hereto, together with the goodwill of the business associated therewith (collectively, the "Marks").

**NOW, THEREFORE**, in consideration of the Purchase Agreement and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

- 1. The Assignor hereby irrevocably assigns to the Assignee: (i) all of the Assignor's right, title, and interest in, to and under the Marks, together with the goodwill of the business associated therewith; (ii) any and all legal actions and rights and remedies at law or in equity for past infringements, misappropriations, or other violations of the Marks, including the right to sue for, collect, and retain all damages, profits, proceeds, and all other remedies associated therewith; and (iii) any and all income, royalties, and payments accruing on or after the Effective Date with respect to the Marks, for the Assignee's own use and enjoyment and for the use and enjoyment of the Assignee's successors, assigns, or other legal representatives.
- 2. The Assignor agrees not to use (and to terminate and discontinue all use of) the terms contained in the Marks and any terms similar thereto in any domain name, domain name registration, trademark, service mark, trade name, company name, legal name, fictitious business name, logo, and any other indicator of origin, with regard to itself and any business or entity at any time owned or controlled by the Assignor.
- 3. The Assignor authorizes and requests the Commissioner of Patents and Trademarks of the United States, and any other official throughout the world whose duty is to register and record ownership in trademark registrations and applications for registration of trademarks, to record the Assignee as the assignee and owner of any and all of the Assignor's rights in the Marks.
- 4. This Agreement is executed and delivered pursuant to the Purchase Agreement. This Agreement is subject in all respects to the terms and conditions of the Purchase Agreement,

04/23/2018 **201707468.7** 

and all of the representations, warranties, covenants, agreements and limitations of the Assignor and the Assignee contained therein, all of which shall survive the execution and delivery of this Agreement in accordance with the terms of the Purchase Agreement. Nothing contained in this Agreement shall be deemed to supersede, enlarge or modify any of the obligations, agreements, covenants or warranties of the Assignor and the Assignee contained in the Purchase Agreement. Notwithstanding anything to the contrary contained in this Agreement, in the event of any conflict between the terms of this Agreement and the terms of the Purchase Agreement, the terms of the Purchase Agreement shall control.

- 5. All of the terms and provisions of this Agreement shall inure to the benefit of and be binding upon the parties hereto and their respective successors and assigns.
- 6. No provision of this Agreement is intended to confer upon any Person other than the parties hereto and their respective successors and assigns any rights or remedies hereunder.
- 7. In its entirety Section 7.3 (Governing Law; Consent to Jurisdiction) of the Purchase Agreement is incorporated herein by reference.
- 8. The Assignor agrees to reasonably cooperate with the Assignee in transferring all rights to the Marks to the Assignee and to execute all lawful documents, and make all rightful oaths and declarations, which may be reasonably required, all at the sole cost and expense of the Assignee, but at no remuneration to the Assignee in recordable form, necessary to vest title to the Marks and all other rights transferred and assigned by this Agreement.
- 9. This Agreement may be executed in counterparts, each of which shall be deemed an original, but all of which together shall be deemed to be one and the same agreement. A signed copy of this Agreement delivered by facsimile, e-mail or other means of electronic transmission shall be deemed to have the same legal effect as delivery of an original signed copy of this Agreement.

[REMAINDER OF PAGE INTENTIONALLY LEFT BLANK]

2-

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the date first above written.

**ASSIGNOR:** 

C

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the date first above written.

ASSIGNUR:
STEEP & BREW, INC.
Mark Ballering, President
ASSIGNEE:
GENERATIONS COFFEE COMPANY, LLC
GENERATIONS COFFEE COMITANT, LLC
Day of the state o
By: Name: Andrew Gordon
Title: Manager
By:
Name: Dominic Caruso
Title: Manager

IN WITNESS WHEREOF, the parties hereto have duly executed this Agreement as of the date first above written.

ASSIGNOR:
STEEP & BREW, INC.
Mark Ballering, President
ASSIGNEE:
GENERATIONS COFFEE COMPANY, LLC
n
By: Name: Andrew Gordon
Title: Manager
By: Owner arms
Name: Dominic Caruso
Title: Manager

[Signature Page to Assignment of Trademarks]

# **SCHEDULE A**

#### Marks

# Registered Trademarks:

- 1. Steep & Brew (Reg. No. 1,553,130)
- 2. Whole Bean Steep & Brew Coffees (Reg. No. 1,971,360)
- 3. Whole Bean Steep & Brew Coffees (Reg. No. 1,968,491)
- 4. World Cup (Reg. No. 2,078,262)
- 5. World Cup (Reg. No. 2,076,657)
- 6. World Cup Hawaii Mexico Guatemala El Salvador Columbia Cost Rica Venezuela Brazil Yemen Ethiopia Kenya Tanzania Zimbabwe Java Sumatra New Guinea Crafted by Steep & Brew (Reg. No. 2,706,420)
- 7. Kendra (Reg. No. 4,522,990)
- 8. Café Fair (Reg. No. 4,297,249)
- 9. K2 Coffee (Reg. No. 4,502,625)
- 10. Club Fund Zone (Reg. No. 4,086,431)
- 11. Coffee Done Right (Reg. No. 3,816,076)
- 12. Coffee Done Right (Reg. No. 3,812,456)
- 13. Toasted Nut (Reg. No. 2,774,681)
- 14. Toasted Nut (Reg. No. 1,955,449)
- 15. We Know Beans About Coffee (Reg. No. 1,960,051)
- 16. We Know Beans About Coffee (Reg. No. 1,892,293)
- 17. Coffee Craftsmen (Reg. No. 2,082,016)
- 18. Coffee Craftsmen (Reg. No. 2,051,356)

#### Unregistered Trade Name:

1. Butlers' Brew

RECORDED: 03/04/2019

# EXHIBIT 3

# 900819197 12/06/2023

# TRADEMARK ASSIGNMENT COVER SHEET

Electronic Version v1.1 Stylesheet Version v1.2 ETAS ID: TM858891

SUBMISSION TYPE:	RESUBMISSION		
NATURE OF CONVEYANCE:	ASSIGNMENT OF THE ENTIRE INTEREST AND THE GOODWILL		
RESUBMIT DOCUMENT ID:	900816603		

# **CONVEYING PARTY DATA**

Name	Formerly	Execution Date	Entity Type
Generations Coffee Company, LLC		11/27/2023	Limited Liability Company: DELAWARE

#### **RECEIVING PARTY DATA**

Name:	Express Coffee LLC	
Street Address:	N28W23000 Roundy Drive, Suite 202	
City:	Pewaukee	
State/Country:	WISCONSIN	
Postal Code:	53072	
Entity Type:	Limited Liability Company: WISCONSIN	

#### **PROPERTY NUMBERS Total: 8**

Property Type	Number	Word Mark
Registration Number:	1553130	STEEP & BREW
Registration Number:	1968491	WHOLE BEAN STEEP & BREW COFFEES
Registration Number:	1971360	WHOLE BEAN STEEP & BREW COFFEES
Registration Number:	3812456	COFFEE DONE RIGHT
Registration Number:	3816076	COFFEE DONE RIGHT
Registration Number:	4297249	CAFÉ FAIR
Registration Number:	4522990	KENDRA
Serial Number:	97826634	STEEP AND BREW

#### **CORRESPONDENCE DATA**

**Fax Number:** 2125894201

Correspondence will be sent to the e-mail address first; if that is unsuccessful, it will be sent

using a fax number, if provided; if that is unsuccessful, it will be sent via US Mail.

**Phone:** 2125894200

Email: bhipdocket@bakerlaw.com
Correspondent Name: Robert B.G. Horowitz, Esq.
Address Line 1: 45 Rockefeller Plaza, 14th Floor

Address Line 2: Baker & Hostetler LLP

Address Line 4: New York, NEW YORK 10111

TRADEMARK REEL: 008317 FRAME: 0609

900819197

ATTORNEY DOCKET NUMBER:	129386.000001	
NAME OF SUBMITTER:	Robert B.G. Horowitz	
SIGNATURE:	/Robert B.G. Horowitz/	
DATE SIGNED:	12/06/2023	
Total Attachments: 2		

source=Assignment (Signed)#page1.tif source=Assignment (Signed)#page2.tif

# TRADEMARK ASSIGNMENT

Whereas, Generations Coffee Company, LLC, a Delaware limited liability company having its place of business at 850 New Burton Road, Dover, Delaware 19904 owns the following trademark registrations and pending trademark application:

# A. Registrations

No. 1968491 for WHOLE BEAN STEEP & BREW COFFEES (logo)

No. 1971360 for WHOLE BEAN STEEP & BREW COFFEES (logo)

No. 3816076 for COFFEE DONE RIGHT

No. 3812456 for COFFEE DONE RIGHT

No. 4522990 for KENDRA

No. 4297249 for CAFÉ FAIR

No. 1553130 for STEEP & BREW

# B. Application

SN 97826634 for STEEP AND BREW

Whereas, Express Coffee LLC, a Wisconsin limited liability company having its place of business at N28W23000 Roundy Drive Suite 202, Pewaukee, Wisconsin 53072 desires to acquire all rights, title and interests in and to the aforesaid marks and registrations and pending application therefor:

Now, therefore, for \$10.00 and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, said Generations Coffee Company, LLC does hereby assign unto said Express Coffee LLC all rights, title and interests in and to the aforesaid marks, together with all goodwill of the business symbolized by the marks, and the trademark registrations and pending application

therefore, none of which are subject to any claims of any kind, proceedings of any kind, licenses, liens, pledges, or encumbrances of any kind.

Generations Coffee Company, LLC

Date: November 27,2023

Dover, Delaware

BY: PRES IDENT FROM CULL

Andrew Gordon, Managing Member

TRADEMARK REEL: 008317 FRAME: 0612

RECORDED: 01/26/2028